

# The Edinburgh Gazette

Published by Authority

# FRIDAY 1st AUGUST 1986

Registered as a Newspaper at the Post Office

The following Notice was taken from *The London Gazette* of Tuesday, 29th July 1986.

# TREASURY

Treasury Chambers, London S.W.1.

25th July 1986

### TENDERS FOR TREASURY BILLS

- 1. THE Lords Commissioners of Her Majesty's Treasury hereby give notice that Tenders will be received at the Securities Office, Bank of England, Threadneedle Street, on Friday, 1st August 1986, at 1 p.m., for Treasury Bills to be issued under the Treasury Bills Act 1877, and the National Loans Act 1968, to the amount of £100,000,000.
- The Bills will be in amounts of £5,000, £10,000, £25,000, £50,000, £100,000, £250,000 or £1,000,000. They will be dated at the option of the tenderer on any business day from Monday, 4th August 1986 to Friday, 8th August 1986 inclusive and will be due 91 days after date.
- 3. The Bills will be issued and paid at the Bank of England.
- 4. Each Tender must be for an amount of not less than £50,000 and must specify the date on which the Bills required are to be dated and the net amount per cent (being a multiple of one halfpenny) which will be given for the amount applied for. Separate Tenders must be lodged for Bills of different dates.
- Tenders must be made through a London Banker, Discount House or Broker.
- 6. Notification will be sent on the same day as Tenders are received to the persons whose Tenders are accepted in whole or in part. Payment in full of the amounts due in respect of such accepted Tenders must be made to the Bank of England by means of cash or by draft or cheque drawn on the Bank of England not later than 1.30 p.m. on the day on which the relative Bills are to be dated.

- 7. Tenders must be made on the printed forms which may be obtained from the Bank of England.
- The Lords Commissioners of Her Majesty's Treasury reserve the right of rejecting any Tenders.

# THE COMPANIES ACT 1985

NOTICE is hereby given that the name of J. & A. HARGAN LIMITED was inadvertently included in the list of Companies struck off the Register pursuant to Section 652(5) of the Companies Act 1985 (Chapter 6) by notice in the Edinburgh Gazette of 15th July 1986, and the Registrar has been advised that such notice was ineffective to dissolve the Company.

> E.T.K. Lougheed, Registrar of Companies.

Companies Registration Office, 102 George Street, Edinburgh EH2 3DJ. 1st August 1986.

(23)

NOTICE is hereby given pursuant to Section 653 of the Companies Act 1985 that the undernoted Company has been restored to the Register of Companies:

> Rangedon Limited Glasgow Sheriff Court

> > E. T. K. Lougheed, Registrar of Companies.

# **BUILDING SOCIETIES**

# THE BUILDING SOCIETIES

### (AUTHORISED BANKS)

### (No 192) ORDER 1986

THE Chief Registrar of Friendly Societies, pursuant to the powers conferred on him by Section 59(2) of the Building Societies Act 1962, and with the consent of the Treasury, hereby orders as follows:

1. The following banks are hereby designated as authorised to hold funds of building societies as mentioned in Section 59(1) of the Building Societies Act 1962:

Banque Nationale de Paris SA London Branch Citicorp Investment Bank Limited King and Shaxson Limited Postipankki (UK) Limited Republic National Bank of New York

2. This Order may be cited as the Building Societies (Authorised Banks) (No 192) Order 1986.

J M Bridgeman, Chief Registrar of Friendly Societies.

Dated 7th July 1986.

We consent to this Order.

Tim Sainsbury, Michael Neubert, Two of the Lords Commissioners of Her Majesty's Treasury.

Dated 18th July 1986.

(4)

### **BUILDING SOCIETIES**

### THE BUILDING SOCIETIES

## (AUTHORISED BANKS)

### (No 193) ORDER 1986

THE Chief Registrar of Friendly Societies, pursuant to the powers conferred on him by Section 59 of the Building Societies Act 1962, and to all other powers enabling him in that behalf, and with the consent of the Treasury, hereby orders as follows:

- The Building Societies (Authorised Banks) (No 160) Order 1983, which designated Chase Bank (Ireland) Limited as authorised to hold funds of building societies, is hereby revoked.
- The Building Societies (Authorised Banks) (No 82) Order 1976, which designated King & Shaxson Limited (which subsequently changed its name to King & Shaxson plc) as authorised to hold funds of building societies is hereby revoked.
- 3. This Order may be cited as the Building Societies (Authorised Banks) (No 193) Order 1986.

J M Bridgeman, Chief Registrar of Friendly Societies.

Dated 7th July 1986.

We consent to this Order.

Tim Sainsbury, Michael Neubert, Two of the Lords Commissioners of Her Majesty's Treasury.

Dated 18th July 1986.

(5)

### **BUILDING SOCIETIES**

### NOTICE OF CHANGES OF NAMES

THE following banks changed their names during the period 1st April 1986 to 9th July 1986:

- Barclays Merchant Bank Limited (formerly Barclays Bank (France) Limited) authorised by the Building Societies (Authorised Banks) (No 27) Order 1968 has become Barclays de Zoete Wedd Limited.
- Charterhouse Japhet plc (formerly S Japhet & Co Limited) authorised by the Building Societies (Authorised Banks) (No 10) Order 1962 has become Charterhouse Bank Limited.
- County Bank Limited authorised by the Building Societies (Authorised Banks) (No 20) Order 1965 has become NatWest Investment Bank Limited.
- 4. Euro-Latinamerican Bank Limited authorised by the Building Societies (Authorised Banks) (No 190) Order 1986 has become Euro-Latinamerican Bank plc.

(6)

# CONTROL OF POLLUTION ACT 1974 SECTION 36(1)(a)

# NOTICE OF APPLICATIONS FOR CONSENT IN PURSUANCE OF SECTION 34

NOTICE is hereby given in accordance with Section 36(1)(a) of the Control of Pollution Act 1974 that the following applications for consent to discharge have been made to the Clyde River Purification Board:

- Ref. No. D414 A.M. Davidson, "The Pines", Ballat Crossroads, Balfron Station, near Glasgow — discharge of 36 cubic metres/day of sewage effluent to a tributary of the Endrick Water at National Grid Reference NS 524 897 from "The Pines", Ballat Crossroads.
- Ref. No. D421 Strathclyde Regional Council, Department of Sewerage, Renfrew Division, Regional Offices, Cotton Street, Paisley PA1 1LJ — discharge of 2.0 cubic metres/second of sewage effluent to the Firth of Clyde at National Grid Reference NS 250 790 from the sewerage system serving Greenock, Gourock and Port Glasgow.

Any person who wishes to make representations about either of these applications should do so, in writing, to the Clyde River Purification Board, Rivers House, Murray Road, East Kilbride, Glasgow G75 0LA, not later than 12th September 1986, quoting the reference number.

Copies of the respective applications may be inspected free of charge at the above offices of the Board at all reasonable hours.

> D. Hammerton, Director.

Clyde River Purification Board. Date 25th July 1986.

(12)

# MINISTRY OF AGRICULTURE, FISHERIES AND FOOD

# DEPARTMENT OF AGRICULTURE AND FISHERIES FOR SCOTLAND

# WELSH OFFICE, AGRICULTURE DEPARTMENT

### ANIMAL HEALTH ACT 1981

NOTICE is hereby given in pursuance of Section 91(1) of the Animal Health Act 1981 that the Minister of Agriculture, Fisheries and Food and the Secretary of State for Scotland and the Secretary of State for Wales, acting jointly, have made the following Order:

THE SHEEP SCAB ORDER 1986 (S.I. 1986 No. 862)

This order which came into operation on 3rd June 1986 consolidates, with amendments, the provisions of the Sheep Scab Order 1977, and its amending instruments, which are revoked.

The main provisions of the order are as follows:

- Part II contains provisions relating to cases of sheep scab and suspected sheep scab and includes provisions concerning —
  - (a) the compulsory notification of sheep scab (Article 3);
  - (b) the investigation by a veterinary inspector into the existence of sheep scab (Article 6);
  - (c) the declaration as an infected place or places where sheep scab exists or is suspected to exist (Article 5); and
  - (d) rules to be observed in infected places (Article 7).
- (2) Part III contains provisions relating to the movement of sheep and includes provisions which
  - (a) enable restrictions to be imposed on the movement of sheep where this may give rise to the risk of the spread of sheep scab (Article 10);
  - (b) empower an inspector, by notice, to require sheep which are affected or suspected of being affected with sheep scab to be removed from markets or places of exhibition (Article 11); and
  - (c) prohibit the movement of sheep which are affected or suspected of being affected with sheep scab on to, and their grazing on, any common or unenclosed land, unfenced fields or highway, and provides for the seizure, detention and slaughter of sheep moved or grazed contrary to this prohibition (Article 12).
- (3) Part IV contains provisions which apply in areas (called infected areas in the order) which are declared by special order of the Minister of Agriculture, Fisheries and Food to be areas infected with sheep scab to which the provisions of this part of the order apply. Part IV includes provisions concerning —
  - (a) the prohibition of the movement of sheep out of an infected area except under a licence issued by the local authority which will not be issued unless an appropriate declaration has been made by the owner of the sheep or his agent concerning their proposed movement (Article 15);
  - (b) the prohibition of the movement of sheep from any farm, holding or agricultural premises in an affected area to any other farm, holding or agricultural premises subject to certain exceptions (Article 16);
  - (c) the control of the movement of sheep in to and out of markets or places of exhibition in an infected area and of the holding of markets or exhibitions of sheep in such an area (Article 17);
  - (d) the prohibition of the public sale of sheep on farms or infected areas and of the movement of sheep on to such farms for public sale there subject to certain exceptions (Article 18);
  - (e) the compulsory dipping of sheep in an infected area (Article 20);
  - (f) the prohibition of the movement of sheep which have not been dipped as required by the order (Article 23).

- (4) Part V contains general and supplementary provisions and includes provisions which
  - (a) empower an inspector, by notice, to require the owner of sheep which are kept on any common or unenclosed land either to move them off such land and not to move them back on to the land without an inspector's written approval or to gather them together on some specified part of the land and not to move them from there without such approval (Article 26);
  - (b) require a sheep dealer (as defined in the order) to make a record in a specified form relating to every purchase and sale of sheep by him and to keep such record for a period of 12 months from the date of the purchase or sale to which it relates (Article 27);
  - (c) require a person who has dipped any sheep in accordance with a requirement contained in the order to provide a sample of the sheep dip or dipping bath used or to permit an inspector or a constable to take such a sample (Article 30);
  - (d) require the cleansing and disinfection of premises used for dipping sheep and of vehicles used for the carriage of sheep which are affected or suspected of being affected with sheep scab (Articles 32 and 33); and
  - (e) require an auctioneer to make a record relating to every sale of sheep by him and to retain such record for a period of 12 months from the date of the sale to which the record relates (Article 35).
- (5) The changes of substance are -
  - (a) all the provisions of the previous orders requiring sheep to be dipped within 28 days of a specified occurrence are amended so as to require the dipping to have taken place within the previous 21 days (Articles 13, 15, 16, 17 and 18);
  - (b) sheep dealers are required to keep records relating to the purchase and sale of sheep by them (Article 27); and
  - (c) the making of a false declaration for the purposes of the order is declared to be an offence against the Animal Health Act 1981 (Article 41(b)).

Copies of the above mentioned Order may be obtained from Her Majesty's Stationery Office, 49 High Holborn, London WC1V 6HB, or 13A Castle Street, Edinburgh EH2 3AR, price £4.90p.

(13)

(77)

AVERAGE prices of British corn published pursuant to the Corn Returns Act 1882 as amended. Prices represent the average for all sales during the week ended 17th July 1986.

BRITISH CORN	Average price in pounds per tonne
Wheat	103.7
Barley	99.3
Oats	99.6
DAFS,	
Chesser House,	
Gorgie Road,	
Edinburgh EH11 3AW.	

### DEPARTMENT OF EMPLOYMENT

# SEX DISCRIMINATION ACT

# A NOTICE OF AN ORDER MADE UNDER SECTION 47(4)(b) OF THE SEX DISCRIMINATION ACT 1975

THE Secretary of State on 23rd July 1986 designated Edinburgh Women's Training Centre Limited as a training body for the purposes of subsections (1) and (2) of Section 47 of the Sex Discrimination Act 1975.

Sandra Webber.

25th July 1986.

# DEPARTMENT OF ENERGY

### THE PETROLEUM (PRODUCTION) ACT 1934

THE Secretary of State for Energy hereby gives notice that in pursuance of the powers conferred by Section 2(1) of the Petroleum (Production) Act 1934 and in accordance with the Petroleum (Production) (Landward Areas) Regulations 1984 he has granted 12 Exploration Licences (EXLs) for areas in Scotland. The names of the licensees and the areas covered by the new licences executed to date and not previously recorded in the Gazette are shown on the schedule below. Except where otherwise indicated, the areas are bounded by straight lines drawn between National Grid reference points.

Licence No.	Name of Licensee		County/ Regions	Grid Points of Boundaries
EXL 021	Industrial Scotland Energy plc Hunt Oil (UK) Limited Tricentrol Exploration UK Limited Bow Valley Petroleum Limited	538.29	Highland Region of Scotland	ND0000 6886 thence by a line following LWMOST to ND0200 7000-ND1008 7000 thence by a line following LWMOST to ND1362 7000- ND1708 7000 thence by a line following LWMOST to ND2000 7196-ND2000 7672 thence by a line following LWMOST to ND3000 7488- ND3000 5000-ND1000 5000 ND1000 6000-ND0000 6000- ND0000 6886
EXL 025	Burmah Oil Exploration Limited Berkeley Resources Limited Attock Oil Exploration (UK) Limited Saxon Oil plc	<b>52.00</b>	Fife and Tayside Regions of Scotland	NO2000 0000-NO2200 0000- NT2200 9400-NT3000 9400- NT3000 9000-NT2000 9000- NO2000 0000
				(8)

### PRIVY COUNCIL OFFICE

### **ARCHITECTS (REGISTRATION) ACT 1931**

THE following amending Regulations made by the Architects Registration Council of the United Kingdom under Section 13 of the above-mentioned Act have been submitted for the approval of the Privy Council.

### REGULATIONS

At end of Regulation 1(1) insert:-

"" prescribed manner" for the purpose of paragraph 2 of Schedule 2 to the Act means a notice in writing signed by the Chairman, the Vice-Chairman or the Clerk of the Board of Architectural Education, addressed to the Council, and containing particulars of a person nominated for membership of the Board in accordance with the provisions of the said paragraph, and of the school of architecture the governing body of which has recommended that person for such membership.

"prescribed number" for the purpose of paragraph 2 of Schedule 2 to the Act shall be twenty five".

The purpose of these Regulations is to prescribe the procedure for enlarging the Board of Architectural Education with nominees of additional schools of architecture and to prescribe to a maximum number of 25 such nominees.

Representations by persons to whom the Regulations are applicable may be made in writing to the Privy Council on or before the 15th day of September 1986.

(16)

# RATES OF INTEREST ON LOANS BY THE PUBLIC WORKS LOAN COMMISSIONERS TO LOCAL AUTHORITIES

#### NOTICE

### TREASURY MINUTE dated 30th July 1986

THE Lords Commissioners of Her Majesty's Treasury read Section 3(2) of the National Loans Act 1968 (1968 C.13 as amended) whereby any sums borrowed from the Public Works Loan Commissioners shall bear interest at such rates as the Treasury may determine from time to time in accordance with Section 5 of the said Act.

My Lords understand that the Public Works Loan Commissioners will be operating arrangements under which local authorities are permitted to borrow from the Commissioners up to a specified quota at a certain fixed rate of interest and in excess of such quota at one of two higher fixed rates.

In accordance therewith the Chancellor of the Exchequer now recommends that with effect from 30th July 1986 and until the coming into operation of a further determination:

- I. The rate of interest applicable to any such fixed rate loan will be that in force on the day the authority applies to borrow or, if the application is before noon three working days before the loan is credited to the account of the authority's banker, the rate in force at 11 a.m. two banking days before credit day;
- II. Different rates of interest shall be charged on such fixed rate loans according to whether the principal of a loan is repaid by instalments or at maturity and, if repaid by instalments, whether by equal instalments of the principal with interest paid on the decreasing balance of the principal (E.I.P.), or by instalments of equal repayments of the principal and interest paid thereon (E.R.);
- III. The rates of such interest shall be:-

	Per Cent per Annum Loans Repayable		
	By Instalments		At Maturity
P.W.L.B. Quota Rates	E.I.P.	<i>E.R</i> .	
1 year	_	_	10
Over 1 but not over 2 years	10	97/8	93/4
Over 2 but not over 3 years	93/4	93/4	95/8
Over 3 but not over 4 years	93/4	93/4	95/8
Over 4 but not over 5 years	95/8	95/8	95/8
Over 5 but not over 6 years	95/8	95/8	9 5/8
Over 6 but not over 7 years	95/8	95/8	95/8
Over 7 but not over 8 years	95/8	95/8	93/4
Over 8 but not over 9 years	95/8	95/8	93/4
Over 9 but not over 10 years	95/8	95/8	93/4
Over 10 but not over 15 years	93/4	93/4	93/4
Over 15 but not over 25 years	93/4	93/4	95/8
Over 25 years	95/8	95/8	95/8
P.W.L.B. Non-quota A Rates.			
1 year	_	_	103/4
Over 1 but not over 2 years	11	107/8	101/2
Over 2 but not over 3 years	103/4	103/4	103/8
Over 3 but not over 4 years	103/4	103/4	103/8
Over 4 but not over 5 years	105/8	105/8	103/8

\_\_\_\_\_

101/8	101/8	101/8
101/8	101/8	101/8
101/8	101/8	101/4
101/8	101/8	101/4
101/8	101/8	101/4
101/4	101/4	101/4
101/4	101/4	101/8
101/8	101/8	101/8
	101/8 101/8 101/8 101/8 101/8 101/4	101/8 101/8   101/8 101/8   101/8 101/8   101/8 101/8   101/4 101/4

PWLB Non-quota B Rate loans will bear interest at 1 per cent above the corresponding Non-quota A rates.

The amount which a local authority borrows within its annual quota will bear interest at the appropriate rate in the quota set of rates. Authorities may borrow further sums at quota rates at the discretion of the Public Works Loan Commissioners. Other borrowing beyond the quota entitlement will be at the appropriate rate in the non-quota set of rates.

### My Lords concur.

The Treasury determine the rates of interest accordingly.

Treasury Chambers, Parliament Street, London SW1P 3AG. 30th July 1986.

# (49)

(48)

# RATES OF INTEREST ON LOANS FROM THE NATIONAL LOANS FUND

### NOTICE

THE Treasury in pursuance of Section 5 of the National Loans Act 1968 (as amended) hereby give notice that on or after 30th July 1986:---

- I. Different rates of interest shall apply according to whether the principal of a loan is repaid by instalments or at maturity and, if repaid by instalments, whether by equal instalments of the principal with interest paid on the decreasing balance of the principal (E.I.P.), or by instalments of equal repayments of the principal and interest paid thereon (E.R.);
- II. The lowest rates of interest satisfying the conditions laid down in Subsection (3) of the said Section 5 shall be:—

	Per Cent per Annum Loans Repayable		
	By Instalments		At Maturity
	E.I.P.	<i>E.R</i> .	
Up to 1 year	_	—	10
Over 1 but not over 5 years	95/8	95/8	95/8
Over 5 but not over 10 years	95/8	95/8	93/4
Over 10 but not over 15 years	93/4	93/4	93/4
Over 15 but not over 25 years	93/4	93/4	95/8
Over 25 years	95/8	95/8	95/8

Treasury Chambers, Parliament Street, London SW1P 3AG. 30th July 1986.

### ROADS (SCOTLAND) ACT 1984

### THE PERTH-ABERDEEN-INVERNESS TRUNK ROAD (A92)

# (COOKSTON ROAD (CI3K), PORTLETHEN SIDE ROADS) ORDER 198

NOTICE is hereby given that after carefully considering objections to the above draft Order published on 20th December 1985, the Secretary of State has decided not to proceed with the stopping up of Cookston Road, at its junction with the Perth-Aberdeen-Inverness Trunk Road (A92). Alternative proposals, discussed with Grampian Regional Council, to allow one-way access only from the southbound carriageway of the trunk road to Cookston Road will be published shortly.

> M. J. P. Cunliffe, Assistant Secretary.

Scottish Development Department, New St. Andrew's House, Edinburgh.

28th July 1986.

# THE TRUNK ROADS (ROUTE A80) (CASTLECARY-HAGGS) (TEMPORARY CLOSURE) ORDER 1986

THE Secretary of State gives notice that he proposes to make the above Order temporarily closing:-

- (a) that length of the northbound carriageway of the Glasgow-Stirling trunk road (Routes A80/M80) from a point on the A80 615 metres south-west of the south face of the bridge carrying the B816 Bonnybridge-Castlecary Road over the A80 to a point on the M80, 474 metres north-east of the north face of the bridge carrying the A803 Dennyloanhead-Kilsyth Road over the trunk road, a distance of 1924 metres (contraflow will operate on the southbound carriageway),
- (b) the northbound slip road at the Castlecary junction from its junction with the B816 Bonnybridge-Castlecary Road 89 metres in a north-easterly direction to its junction with the A80 trunk road (alternative route via B816 and A8011 joining A80 trunk road at the A8011 junction, Cumbernauld), and
- (c) the northbound slip off the A80 trunk road at the Haggs junction in a westerly then southerly direction for a distance of 259 metres (alternative route leave A80 trunk road at A8011 junction, Cumbernauld and proceed via B816 to Bonnybridge to join A803).

The Order which is required to allow essential maintenance work to be carried out will operate from 16th August until 8th November 1986.

Restrictions will only apply in the immediate vicinity of the works and where indicated by the appropriate traffic signs.

> M.J.P. Cunliffe, Assistant Secretary.

Scottish Development Department.

**(**1**)** 

# THE TRUNK ROADS (ROUTE M9) (POLMONT) (TEMPORARY CLOSURE) ORDER 1986

THE Secretary of State gives notice that he proposes to make the above Order temporarily closing:-

- (a) the westbound carriageway of the Edinburgh-Stirling Motorway (Route M9) at Polmont from a point 418 metres east of the east face of the east bridge carrying the A801 Polmont-Armadale road over the motorway at Lathallan junction in a westerly direction for a distance of 2251 metres (contra-flow on eastbound carriageway), and
- (b) the westbound slip on at Lathallan junction from its junction with the A801 road in a westerly direction for a distance of 773 metres to its junction with the M9 motorway (alternative route via A9 through Polmont, A905 to join motorway at Earlsgate junction, Grangemouth).

The Order which is required to allow essential maintenance work to be carried out will operate from 16th August until 8th November 1986.

The restrictions will only apply in the immediate vicinity of the works and where indicated by the appropriate traffic signs.

> M.J.P. Cunliffe, Assistant Secretary.

Scottish Development Department.

(44)

(2)

### CITY OF ABERDEEN

# TOWN AND COUNTRY PLANNING (LISTED BUILDINGS AND BUILDINGS IN CONSERVATION AREAS) (SCOTLAND) REGULATIONS 1957-77

NOTICE is hereby given that applications for Listed Building Consent and for Planning Permission, with respect to the undernoted subjects, have been submitted to the City of Aberdeen District Council.

The applications and relative plans are available for inspection at the Department of Planning and Building Control, St. Nicholas House, Aberdeen, during normal office hours, and any representations in connection therewith should be made in writing, to the Director of Law and Administration, Town House, Aberdeen AB9 1AQ, within 21 days of this advertisement.

40 Castle Street (Category 'B' building within Conservation Area 2)

9 Albyn Place (Category 'B' building within Conservation Area 4) of a night safe.

Alterations to frontage by the installation

Alterations and erection of extension to provide conservatory and function room.

Jamus J. K. Smith, Director of Law and Administration.

(39)

### ARGYLL AND BUTE DISTRICT COUNCIL

### TOWN AND COUNTRY PLANNING (SCOTLAND) ACTS 1972/77

# TOWN AND COUNTRY PLANNING (GENERAL DEVELOPMENT) (SCOTLAND) ORDERS 1981 TO 1984

NOTICE is hereby given that the undernoted Planning Application has been received by Argyll and Bute District Council and is hereby advertised in accordance with the provisions of the Town and Country Planning (Scotland) Acts and Related Legislation for public inspection and submission of any representations.

Any person who wishes to make representations to the District Council about the application should make them in writing within the period listed below, from the date of publication of this Notice, to the Director of Administration, Kilmory, Lochgilphead.

Copies of the plans and documents submitted with this application may be inspected during all reasonable hours at the location as shown in the Schedule below, during the undernoted period beginning with the date of publication of this Notice.

### SCHEDULE OF PROPOSED DEVELOPMENTS

Applicant/Development Details & Location/ Reference No. D. MacGillivray. Alterations at Pennygown House, Aros, Isle of Mull. 418/86 (LB). Location where plans/documents are available, reason for advertisement, and period within

which comments can be made The Post Office, Aros, Isle of Mull.

Requires Listed Building Consent. (21 days).

James A. McLellan, Director of Administration.

Kilmory, Lochgilphead. Date 31st July 1986

(11)

### BANFF AND BUCHAN DISTRICT COUNCIL

### 38 PENNAN, NEW ABERDOUR

# TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

# TOWN AND COUNTRY PLANNING (LISTED BUILDINGS AND BUILDINGS IN CONSERVATION AREAS)

(SCOTLAND) REGULATIONS 1975-77

NOTICE is hereby given that application has been made to Banff and Buchan District Council as Local Planning Authority for Planning and Listed Building Consent for alterations to dwellinghouse at 38 Pennan, New Aberdour.

Plans and documents relating to this application may be inspected during normal office hours at the Department of Planning and Development, Town House, Low Street, Banff, and the Post Office, Pennan, during the 21 days beginning with the date of publication of this Notice.

Any person who wishes to make representation to the above mentioned Council with regard to the proposals should make them in writing within that period to the Director of Planning and Development, Town House, Low Street, Banff.

> Peter Suttie, Director of Planning and Development.

> > (36)

### BANFF AND BUCHAN DISTRICT COUNCIL

### 62 SALTOUN PLACE, FRASERBURGH

### TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

# TOWN AND COUNTRY PLANNING (LISTED BUILDINGS AND BUILDINGS IN CONSERVATION AREAS) (SCOTLAND) REGULATIONS 1975-77

NOTICE is hereby given that application has been made to Banff and Buchan District Council as Local Planning Authority for Planning and Listed Building Consent for proposed porch extension at 62 Saltoun Place, Fraserburgh.

Plans and documents relating to this application may be inspected during normal office hours at the Department of Planning and Development, Town House, Low Street, Banff, and the Department of Technical Services, 55 Mid Street, Fraserburgh, during the 21 days beginning with the date of publication of this Notice.

Any person who wishes to make representation to the above mentioned Council with regard to the proposals should make them in writing within that period to the Director of Planning and Development, Town House, Low Street, Banff.

> Peter Suttie, Director of Planning and Development.

> > (37)

### BANFF AND BUCHAN DISTRICT COUNCIL

# 4 WEST SKENE STREET, MACDUFF

# TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

# TOWN AND COUNTRY PLANNING (LISTED BUILDINGS

# AND BUILDINGS IN CONSERVATION AREAS) (SCOTLAND) REGULATIONS 1975-77

NOTICE is hereby given that application has been made to Banff and Buchan District Council as Local Planning Authority for Planning and Listed Building Consent for installation of double glazing and front door at 4 West Skene Street, Macduff.

Plans and documents relating to this application may be inspected during normal office hours at the Department of Planning and Development, Town House, Low Street, Banff, during the 21 days beginning with the date of publication of this Notice.

Any person who wishes to make representation to the above mentioned Council with regard to the proposals should make them in writing within that period to the Director of Planning and Development, Town House, Low Street, Banff.

> Peter Suttie, Director of Planning and Development.

> > (38)

(53)

### THE BORDERS REGIONAL COUNCIL

### (CHAPEL STREET, SELKIRK)

### (PROHIBITION OF WAITING) ORDER 1986

NOTICE is hereby given that on 21st July 1986 The Borders Regional Council made the above Order under Sections 1(1) (2) and 2(1) (2) and Part III of Schedule 9 of the Road Traffic Regulation Act 1984.

The Order came into effect on 24th July 1986.

Full details of the restrictions are contained in the Order, a copy of which together with a map showing the area involved, may be examined at the Borders Regional Council Headquarters, Newtown St. Boswells, between the hours of 9 a.m. and 4.30 p.m., Monday to Friday, and at Selkirk Public Library during normal Library opening hours.

Any person wishing to question the validity of the Order or any provision which it contains on the grounds that it is not within the powers conferred by the Road Traffic Regulation Act 1984 or on the grounds that any requirement of the Act or of any instrument made under the Act has not been complied with in relation to the Order may, within six weeks from Thursday, 24th July 1986, apply to the Court of Session for that purpose.

	Kenneth J. Clark, Chief Executive.
Headquarters, St. Boswells.	

24th July 1986.

Regional Newtown

### CUMNOCK AND DOON VALLEY DISTRICT COUNCIL

### TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

# TOWN AND COUNTRY PLANNING (LISTED BUILDINGS) AND BUILDINGS IN CONSERVATION AREAS (SCOTLAND) REGULATIONS 1975

NOTICE is hereby given that application has been made to Cumnock and Doon Valley District Council for alteration to existing dwellinghouse and erection of out-building, Auchmannoch House, Sorn, By Mauchline : Mr Robert Adamson.

A copy of the application and plans and other documents submitted may be inspected during office hours at the Planning Department, Cumnock and Doon Valley District Council, Council Offices, Lugar, Cumnock.

Any person who wishes to make representations about the application should do so to the Director of Administration, Cumnock and Doon Valley District Council, Council Offices, Lugar, Cumnock, within 21 days of the date of this advertisement.

Douglas Hemmings, Director of Administration,

(41)

# FLOOD PREVENTION (SCOTLAND) ACT 1961

# THE DUMFRIES AND GALLOWAY REGIONAL COUNCIL

### (KIRKINNER) FLOOD PREVENTION SCHEME 1986

NOTICE is hereby given that the Dumfries and Galloway Regional Council, in exercise of the powers conferred on them by the Flood Prevention (Scotland) Act 1961 propose to make a flood prevention scheme at Kirkinner to be known as the Dumfries and Galloway Regional Council (Kirkinner) Flood Prevention Scheme 1986, the aim of which will be to mitigate the flooding of properties on the south-west side of Main Street, Kirkinner.

A copy of the proposed Scheme together with the relevant plan may be inspected by any person, free of charge, at Kirkinner Post Office, Kirkinner, and at Council Offices, Dumfries, during normal office hours. Any objections to the proposed Scheme should be sent in writing to the Secretary of State for Scotland, Scottish Development Department, Pentland House, 47 Robb's Loan, Edinburgh, before 1st November 1986.

> George M. Sinclair, Director of Administration and Law.

> > (45)

### DUMFRIES AND GALLOWAY REGIONAL COUNCIL

### TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

THE undernoted applications may be inspected during normal office hours at the locations indicated.

Any representations should be made in writing to me at Council Offices, English Street, Dumfries, within the stated period from the date of this Notice.

Address	Development	Plans At	Period for Representations
NEWTON STEV	WART CONSER	VATION AREA	
Church Street/ Jubilee Road, Newton Stewart (LB/21/86)	Demolition of former grain store and garage	District Council Office, 79 Victoria Street Newton Stewart	21 days
LISTED BUILD			
27 North Main Street, Wigtown (LB/22/86)	Replacement of rubberoid tiles with concrete tiles	Public Library, Wigtown Primary School, Wigtown	21 days
LISTED BUILD	ING CONSENT		
5 Lady Street, Brydekirk (LB/24/86)	Replacement of timber windows with aluminium windows	Department of Physical Planning, 15 Ednam Street Annan	21 days
Old Bank House Lochmaben (LB/25/86)	e Internal and external alterations	Local Government Office, High Street, Lochmaben	21 days
66 Main Street Isle of Whithorn (LB/23/86)	Rebuilding of kitchen	Public Library, St. John Street, Whithorn	21 days

A. H. Dobbie, Director of Physical Planning.

Date 1st August 1986.

### THE EDINBURGH GAZETTE 1st AUGUST 1986

### EAST LOTHIAN DISTRICT COUNCIL

### TOWN AND COUNTRY PLANNING (SCOTLAND) ACTS

NOTICE is hereby given that application for Planning Permission/ Listed Building Consent has been made to East Lothian District Council, as District Planning Authority, as detailed in the Schedule hereto.

The applications and plans submitted are open to inspection at Council Buildings, Haddington, during office hours.

Any representations should be made in writing to the undersigned within 21 days of this date.

#### **SCHEDULE**

- 478/86 Outline application for erection of dwellinghouse in garden ground of Pointe Verte, East Links Road, Gullane. E.J. Cherel (Conservation Area).
- 480/86 Erection of 24 house units and 11 car ports at Old Bakery Site, New Street, Musselburgh. — Margaret Blackwood Housing Ass'c Ltd (Conservation Area).
- 485/86 Erection of stone wall at 44 St. Martins Gate, Haddington. — M. Howie (Conservation Area).
- 486/86 Installation of ventilation system at 112 High Street, Dunbar. — W M Cheung (Conservation Area).
- 488/86 Alterations to cottage at 7A Bayswell Road, Dunbar. Mr & Mrs T.D. Barbour (Conservation Area).
- 491/86 Change of use from fish shop to seafood restaurant/cafe at The Harbour, Port Seton. — J.R. McCaig (Conservation Area).
- 498/86 Re-siting of garage at 10 Briery Bank, Haddington. A Maughan (Conservation Area).
- 501/86 Painting of gable rear wall and front window surrounds of building at 15 High Street, Dunbar. E B Simmons (Conservation Area).

Malcolm Duncan, Director of Administration.

Council Buildings, Haddington.

uwwington.

1st August 1986.

(42)

# THE CITY OF EDINBURGH DISTRICT COUNCIL

# TOWN AND COUNTRY PLANNING (SCOTLAND) ACTS 1972/77

IN accordance with the above Acts and Related Legislation, the following applications and the drawings and other documents submitted with them may be examined at the Planning Department, 18 Market Street, between 9 a.m. and 4.45 p.m., Monday to Thursday and between 9 a.m. and 3.30 p.m. Friday. Written comments may be made to the Director of Planning at the above address within the period specified.

Applications involving Listed Buildings; Conservation Areas; Development Plan departures; other matters of particular public interest (written comments to be made within 21 days of the date of this Notice):-

Location	Ref. No.	Nature of Proposals
9-11 Mary's Place	1305/86	Remove conditions of consent restricting hours of operation of restaurant
26 East Mayfield	1365/86	Proposed new fence
8 West Maitland Street	1366/86	Erection of a canopy to restaurant
2 Rutland Piace/ Rutland Street	1368/86	Alterations to casino & proposed stone cleaning
19 Cumberland Street	1370/86	Alterations & rehabilitation of flats

24 Polwarth Terrace	1377/86	Residential development (in outline)
15 Thirlestane Lane	1381/86	Alterations & subdivision of flat & garages to form 2 flats
180-182 High Street	1386/86	Alterations & part change of use to form offices
61A Hailes Gardens	1388/86	Erection of a dwellinghouse
Tower Farm, Liberton	1390/86	Erection of a covered riding school and temporary store
Holyrood/Abbey Breweries Holyrood Road	1391/86	Proposed demolition
Glasgow Road (former Nos. 1-17)	1393/86	Erection of a supermarket & associated car parking
14 Mayfield Gardens	1402/86	Erection of 9 flats & 11 garages

Nicholas Fort, Director of Planning.

(70)

#### TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

# THE STOPPING UP OF ROADS (CITY OF GLASGOW DISTRICT) (BALURE STREET AND LEDAIG PLACE) ORDER 1986

THE City of Glasgow District Council hereby gives notice that it has confirmed an Order under Section 198A of the Town and Country Planning (Scotland) Act 1972, authorising the stopping up of:-

### 1. Part of Balure Street

### 2. Part of Ledaig Place

A copy of the Order and relevant plan specifying the lengths of roads to be stopped up may be inspected at the Planning Department, 231 George Street, Glasgow G1 1RX, by any person, free of charge, at all reasonable hours.

> James H. Rae, Director of Planning.

Dated: 1st August 1986.

(72)

# CITY OF GLASGOW DISTRICT COUNCIL

# PUBLICITY FOR PLANNING AND LISTED BUILDING APPLICATIONS

THE applications listed below, together with the plans and other documents submitted with them, may be examined at the Planning Department, City of Glasgow District Council, 231 George Street, Glasgow Gl 1RX, between the hours of 9 a.m. and 4.30 p.m., Mondays to Fridays.

1170

(<del>4</del>2

Representations regarding any application within the following category should be made in writing to the above address within 21 days.

# TOWN AND COUNTRY PLANNING (LISTED BUILDINGS AND BUILDINGS IN CONSERVATION AREAS) (SCOTLAND) REGULATIONS 1975

App. No.	Address	Proposed Development
01228/86	174 Craighall Road North Speirs Wharf 158-260 North Speirs Wharf Glasgow G4	Conversion of whisky storage bond into 194 houses, car parking, club rooms, puppet theatre, leisure club, workshop, craft/retail units, business centre, gallery and exhibi- tion space, restaurant and wine bar; conversion of ticket office into hotel, restaurant & wine bar and caretakers flat; and erection of 12 mews houses with covered car parking
01456/86	4-16 York Street 116-124 Broomielaw 11-17 Robertson Street Glasgow G2	Demolition of buildings within conservation area
01717/86	52 Charlotte Street Glasgow G1	Conversion of building to residential to form 6 flatted dwelling units with internal and external alterations
01748/86	45 St Vincent Crescent Glasgow G3 8NG	Sub-division of ground and basement floors of tenement property into two self contained flatted dwelling houses
01758/86	106-108 Hope Street Glasgow G2 6PS	Stonecleaning of building fascades
01823/86	48 Carlton Place Glasgow G5 9TW	Erection of non illuminated sign
01835/86	254 Sauchiehall Street Glasgow G2 3EQ	Installation of shop- front, internal alterations, display of individually illuminated letters with 2 projecting box signs

Representations regarding any application within the following category should be made in writing to the above address within 21 days.

### NOTICE UNDER SECTION 25

# (SETTING OF A LISTED BUILDING) OR (DEVELOPMENT IN A CONSERVATION AREA)

01456/86

4-16 York Street Demolition of buildings 116-124 Broomielaw within conservation area 11-17 Robertson Street Glasgow G2

> J.H. Rae, Director of Planning.

> > (71)

### NOTICES UNDER THE PLANNING ACTS

### GORDON DISTRICT

THE following Planning Applications can all be inspected at the Planning Derpartment in Gordon House, Blackhall Road, Inverurie AB5 9WA, and some are also available at the places specified in Column 4 below. Any person wishing to make representations should do so in writing (quoting the Reference Number and stating clearly the reasons for their comments) to the Director of Planning at the above address, within the time period specified.

	widdin die dine j	chod specifica.		
ub	Address	Proposal Reference No.	Applicant	Plans also available at
, iess ibi- and		EQUIRING DING CONSENT ng representations	— 21 days	
f and xtion h	Edgehill, Kemnay Road, Port Elphinstone, Inverurie	Alterations and extension of building (Ref. G/86/P/914)		
is ea	SETTING OF L	FFECTING THE ISTED BUILDIN ng representations	IGS	
i S vith	Lynwood, Chapel of Garioch	Erection of house in garden (outline) (Ref. G/86/P/912)	Mrs E. Walker (as proposal)	
1			<i>E. Geraldin</i> Director of P	
two			·	(1 <b>9</b> )
	H	IIGHLAND REC	JIONAL COUNCIL	
ng	-		NTRY PLANNING	
	LISTED BU	ILDINGS (SCOT	LAND) REGULAT	ONS 1975

NOTICE is hereby given that application has been received from North British Trust Hotels Ltd, for Listed Building Consent in respect of Highland Hotel, Union Road, Fort William.

Details can be seen at the Divisional Planning Office, Tweeddale, Fort William.

Representations may be made to the undersigned on or before 21 days from the date of publication of this Notice.

D.A.B. Blair, Clerk to the Divisional Planning Committee.

Lochaber House, High Street, Fort William.

(18)

# HIGHLAND REGIONAL COUNCIL

# TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

# TOWN AND COUNTRY PLANNING (LISTED BUILDINGS AND BUILDINGS IN CONSERVATION AREAS) (SCOTLAND) REGULATIONS 1975

APPLICATION has been received for the following development — Erection of signs and installation of an automated teller machine within

a reglazed shopfront at Abbey National Building Society, 1 Union Street, Inverness.

Plans can be seen at Inverness Divisional Planning Office, Town House, Inverness.

Representations may be made to G. Boyd, Divisional Planning Officer, Town House, Castle Street, Inverness, on or before 21 days from the date of this publication.

(40)

### KILMARNOCK AND LOUDOUN DISTRICT COUNCIL

# TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

### TOWN AND COUNTRY AMENITIES ACT 1974 PROPOSED CONSERVATION AREA: MAIN STREET, DUNLOP

KILMARNOCK AND LOUDOUN DISTRICT COUNCIL, in pursuance of its powers under Section 262 of the Town and Country Planning (Scotland) Act 1972, as substituted by Section 2(1) of the Town and Country Amenities Act 1974, hereby gives notice that it has decided to cancel the existing Conservation Areas known as Main Street, Dunlop Conservation Area and Main Street, Dunlop Outstanding Conservation Area and to designate a new Conservation Area to be known as Main Street, Dunlop Conservation Area. All properties previously included in the above named existing Conservation Areas are included in the new Conservation Area. This action is taken to combine the two adjacent Conservation Areas in Dunlop.

### DESIGNATION OF CONSERVATION AREA

The proposed Conservation Area will comprise:

Lugton Road — Numbers 2 and 4 Main Street — Numbers 2-10, 14-20, 22A, 22-34, 38, 42-58, 62-92, 3-7, 11-13, 17-23, 29, 33, 35, 39, 43-59, 67, 67A, 69-97. Dunlop Kirk and Cemetery.

Stewarton Road Numbers 1-5, 6.

The new Conservation Area is shown on a map available for public inspection free of charge during reasonable hours at the Planning Service and Information Office, Civic Centre, John Dickie Street, Kilmarnock.

The effect of designation will be to secure greater control over the demolition of unlisted buildings within the Conservation Area. Furthermore, the demolition of an existing building within the designated area shall be an offence under Section 53 of the Town and Country Planning (Scotland) Act 1972 and Article 13 of the Town and Country Planning (Listed Buildings and Conservation Areas) (Scotland) Regulations 1975, unless Listed Building Consent has first been obtained from the District Planning Authority.

# DIVERSION OF ROADS (LIME ROAD, LOUDOUNHILL QUARRY, DARVEL) ORDER 1986

The Kilmarnock and Loudoun District Council, hereby give notice that they intend to make an Order under Section 198A of the Town and Country Planning (Scotland) Act 1972, authorising the diversion of LIME ROAD, LOUDOUNHILL QUARRY, DARVEL.

Copies of the proposed Order and relevant plan, specifying the road to be diverted, may be inspected at the offices of the Planning Service, Civic Centre, Kilmarnock, by any person, free of charge, at all reasonable hours during a period of 28 days from Friday, 1st August 1986. Within that period any person may, by notice to the Kilmarnock and Loudoun District Council, object to the Order.

# STOPPING UP OF ROADS (HIGHBURN PLACE, KILMARNOCK) ORDER 1986

The Kilmarnock and Loudoun District Council hereby give notice that they intend to make an Order under Section 198A of the Town and Country Planning (Scotland) Act 1972, authorising the stopping up of:-

#### HIGHBURN PLACE, KILMARNOCK

Copies of the proposed Order and relevant plan, specifying the roads to be stopped up may be inspected at the offices of the Planning Service, Kilmarnock and Loudoun District Council, by any person free of charge, at all reasonable hours during a period of 28 days from Friday, 1st August 1986. Within that period any person may, by notice to the Kilmarnock and Loudoun District Council, object to the Order.

# STOPPING UP OF ROADS (LIME ROAD, LOUDOUNHILL OUARRY, DARVEL) ORDER 1986

The Kilmarnock and Loudoun District Council hereby give notice that they intend to make an Order under Section 198A of the Town and Country Planning (Scotland) Act 1972, authorising the stopping up of LIME ROAD, LOUDOUNHILL QUARRY, DARVEL.

Copies of the proposed Order and relevant plan, specifying the road to be stopped up, may be inspected at the offices of the Planning Service, Civic Centre, Kilmarnock, by any person, free of charge, at all reasonable hours during a period of 28 days from Friday, 1st August 1986. Within that period any person may, by notice to the Kilmarnock and Loudoun District Council, object to the Order.

# STOPPING UP OF ROADS (KIRKLAND AVENUE, KILMARNOCK) ORDER 1986

The Kilmarnock and Loudoun District Council hereby give notice that they intend to make an Order under Section 198A of the Town and Country Planning (Scotland) Act 1972, authorising the stopping up of:-

# KIRKLAND AVENUE AND HILLBANK ROAD: ALTONHILL, KILMARNOCK

Copies of the proposed Order and relevant plan, specifying the roads to be stopped up, may be inspected at the offices of the Planning Service, Kilmarnock and Loudoun District Council by any person, free of charge, at all reasonable hours during a period of 28 days from Friday, 1st August 1986. Within that period any person may, by notice to the Kilmarnock and Loudoun District Council, object to the Order.

### **SECTION 25**

### TOWN AND COUNTRY PLANNING (LISTED BUILDINGS

AND BUILDINGS IN CONSERVATION AREAS)

NOTICE OF APPLICATION FOR LISTED BUILDING CONSENT

Notice is hereby given that an application is being made to the Council for Listed Building Consent to CARRY OUT ALTERATIONS AT GROUND, FIRST AND SECOND FLOORS TO FORM OFFICES AT 55 JOHN FINNIE STREET, KILMARNOCK.

A copy of the application form and plan may be inspected at the offices of the Planning Service, Civic Centre, John Dickie Street, Kilmarnock.

Any representations about the proposal should be made in writing stating the grounds on which it is made and sent to the undersigned before 22nd August 1986.

James McAndrew, District Administration Manager.

PO Box 13, Civic Centre, Kilmarnock KA1 1BY.

1st August 1986.

(69)

### TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972 -

# TOWN AND COUNTRY PLANNING (LISTED BUILDINGS AND BUILDINGS IN CONSERVATION AREAS) (SCOTLAND) REGULATIONS 1975

NOTICE is hereby given that application is being made to Kyle and Carrick District Council by the undernoted for Planning Permission in respect of the properties named.

Copies of the applications and plans may be inspected at the office of the Director of Architecture, Planning and Technical Services, Burns House, Burns Statue Square, Ayr.

Any person who wishes to make representations about the application should do so to the Director of Administration, Kyle and Carrick District Council, Burns House, Burns Statue Square, Ayr, within twenty-one days of the date of this advertisement.

### Listed Building in Conservation Area

Ethel Austin Investment Properties Limited c/o Mason Owen and Partners Glastone House Union Court Castle Street	Demo and fl of new upper extens retail
Liverpool	Ayr
	Properties Limited c/o Mason Owen and Partners Glastone House Union Court Castle Street

olition of shops lats and erection w retail unit and floor flat and sion to existing unit - 2/6 Hope 1/51 High Street,

Ian R. D. Smillie. Director of Administration.

(10)

# MIDLOTHIAN DISTRICT COUNCIL

THE following applications may be examined at the Department of Planning and Building Control, 17 Westfield Park, Eskbank, Dalkeith, from 9.15 a.m. to 4.45 p.m. Mondays to Thursdays, and from 9.15 a.m. to 3.30 p.m. Fridays.

### Listed Building Consent

Blairesk Hall,	Demolition of roadside	
Polton Road, Loanhead	stone wall	

Please submit any comments in writing to me, not later than 21st August 1986.

> .R.W. Maslin, Director of Planning and Building Control.

> > (34)

### MORAY DISTRICT COUNCIL

# THE TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

### THE MORAY DISTRICT COUNCIL

### (STOPPING UP OF SECTION OF HILL TERRACE, ELGIN) **ORDER 1986**

THE Moray District Council hereby give notice that they have made an Order under Section 198A of the Town and Country Planning (Scotland) Act 1972, authorising the stopping up of a length of Hill Terrace, Elgin, for a distance of 8.5 metres or thereby from its junction with High Street, Elgin.

Copies of the Order and relevant plans specifying the section of roadway to be stopped up may be inspected at the offices of the undersigned by any person free of charge, at all reasonable hours during the period of 28 days from 1st August 1986.

Within that period any person may, in writing to the undersigned, object to the making of the Order.

> James P.C. Bell. Chief Executive.

District Headquarters, Elgin.

1st August 1986.

### MORAY DISTRICT COUNCIL

# TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

# TOWN AND COUNTRY PLANNING (LISTED BUILDINGS AND BUILDINGS IN CONSERVATION AREAS) (SCOTLAND) REGULATIONS 1975

NOTICE is hereby given that application has been made to Moray District Council as local planning authority for Planning Permission and/or Listed Building Consent to:-

Alter boundary walls at Dr. Gray's and Maryhill Hospitals, Elgin.

A copy of the application and plan and other documents submitted with it may be inspected during normal office hours at the office of the Director of Physical Planning and Development, District Headquarters, High Street, Elgin within a period of 21 days following the date of publication of this Notice.

Any person who wishes to make any objections or representations in respect of the application should do so in writing within the aforesaid period to the Chief Executive, District Headquarters, High Street, Elgin.

> James P.C. Bell, Chief Executive

District Headquarters, High Street, Elgin. Dated 1st August 1986.

(33)

### PERTH AND KINROSS DISTRICT COUNCIL

### TOWN AND COUNTRY PLANNING (SCOTLAND) ACTS ETC.

THE undernoted planning applications have been received and are advertised in accordance with the provisions of the Town and Country Planning (Scotland) Acts and Related Legislation, for public inspection and submission of any representations.

A copy of the plans and documents submitted may be inspected during normal office hours, Monday to Friday, at the office of the Director of Planning and Industry, Council Building, 2 High Street, Perth, and at the Local Area Office as shown in the Schedule below, during the 21 days following the date of publication of this Notice.

Any person who wishes to make representations about any of the applications should do so in writing to the Director of Planning and Industry at the above address within 21 days of the date of publication of this Notice.

### SCHEDULE OF PROPOSED DEVELOPMENTS

Development — Location — Developer	Local Area Office and Reason for Publication
Alterations and part change of use of plumber's shop and store into three flats at Haxton's, Cathedral Street, Dunkeld. Haxton's of Dunkeld	Listed Building Consent and development affecting conservation areas
Erect a kitchen and bathroom extension at Allt-Essan, Killiecrankie. Mr. & Mrs. Stebbings.	Environmental Health Office, 26 Atholl Road, Pitlochry. Listed Building Consent
Change of use from shop/house/ basement into consulting rooms	Listed Building Consent

David R. Penman. Director of Planning and Industry.

(68)

at 7 Charlotte Street, Perth. Mr. B. MacKenzie.

### THE EDINBURGH GAZETTE 1st AUGUST 1986

### COMHAIRLE NAN EILEAN

# WESTERN ISLES ISLANDS COUNCIL

### (VARIOUS STREETS, STORNOWAY)

### (PROHIBITION & RESTRICTION OF WAITING) ORDER 1986

- 1. THE Western Isles Islands Council proposes to make an Order under Section 1 of the Road Traffic Regulation Act 1984.
- 2. When this Order comes into effect:
  - (a) vehicles must not wait any time in any of the lengths of roads specified in Schedule 1 to this Notice.
  - (b) vehicles must not wait during the hours of 8 a.m. to 6 p.m. in any of the lengths of road specified in Schedule 2 to this Notice, and
  - (c) vehicles must not wait for more than 30 minutes in any hour between the hours of 8 a.m. and 6 p.m. in any of the lengths of road specified in Schedule 3 to this Notice.
- 3. Exceptions will permit:
  - (a) picking up and setting down of passengers;
  - (b) loading or unloading of goods;
  - (c) where necessary in connection with the maintenance of the roads or adjacent buildings and services;
  - (d) where necessary, in pursuance of statutory powers and duties; and
  - (e) by disabled drivers.
- Current Traffic Regulation Orders insofar as they impose waiting restrictions on any of the lengths of road specified in the Schedules to this Order will be revoked by the proposed Order.
- 5. Full details of these proposals are in the draft Order which, together with a map showing the restricted lengths of road and a statement of the Council's reasons for proposing to make the Order, may be examined at the Council Offices, Sandwick Road, Stornoway between the hours of 9 a.m. and 5.30 p.m. on Mondays to Fridays.
- 6. Any person wishing to object to the proposed Order should send details of the grounds for objection in writing to the Director of Administration, Council Offices, Sandwick Road, Stornoway within 21 days of the date of publication of this Notice.

### SCHEDULE 1:

(no waiting at any time)

South Beach Street on its northern side from the Town Hall to Kenneth Street.

James Street on its northern side from its junction with South Beach Street for 30 metres.

Shell Street on its north eastern side from its junction with James Street for 30 metres.

#### SCHEDULE 2:

(During the hours 8 a.m. to 6 p.m.)

South Beach Street on its southern side from the exit at the Weigh Bridge at No 2 pier to a point opposite the dwelling house at No 3 Shell Street. Shell Street on its north eastern side from No 3 Shell Street to its junction with Ferry Road.

North Beach Street on its northern side from its junction with Bank Street to its junction with Castle Street.

James Street on its southern side from its junction with Keith Street to its junction with Matheson Road.

### Lewis Street

- (a) On its eastern side from its junction with James Street to a point 8 metres from its junction with Francis Street; and
- (b) On its eastern side from a point from its junction with Francis Street to a point 8 metres from its junction with Church Street.

Church Street

- (a) On its northern side from a point 8 metres from its junction with Kenneth Street to a point 8 metres from its junction with Keith Street; and
- (b) On its southern side from a point 8 metres from its junction with Keith Street to a point 8 metres from its junction with Lewis Street.

### Francis Street

- (a) On its northern side from a point 8 metres from its junction with Kenneth Street to a point 8 metres from its junction with Keith Street;
- (b) On its northern side from a point 8 metres from its junction with Keith Street to a point 8 metres from its junction with Lewis Street; and
- (c) On its northern side from a point 8 metres from its junction with Lewis Street for 24 metres eastwards.

### SCHEDULE 3:

(Limited Waiting by Vehicles — 30 minutes in any hour between 8 a.m. and 6 p.m.).

South Beach Street

- (a) On its northern side between its junction with Quay Street to its junction with Cromwell Street; and
- (b) On its northern side between its junctions with Kenneth Street and James Street.

Quay Street on its eastern side between its junctions with South Beach Street and Point Street.

Francis Street

- (a) On its southern side from a point 8 metres from its junction with Kenneth Street to a point 8 metres from its junction with Keith Street; and
- (b) On its southern side from a point 8 metres from its junction with Keith Street to a point 8 metres from its junction with Lewis Street.

Robert J.C. Barnett, Director of Administration.

(43)

# GRAMPIAN REGIONAL COUNCIL ROAD TRAFFIC REGULATION ACT 1984

# THE GRAMPIAN REGIONAL COUNCIL (COOKSTON ROAD, PORTLETHEN) (TRAFFIC MANAGEMENT) ORDER 1986

THE Grampian Regional Council propose to make the above-named Order under Sections 1(1) to (3) and 2(1) to (3) of the Road Traffic Regulation Act 1984. The effect of the Order will be to prohibit the movement of any vehicle on that length of Cookston Road, Portlethen, extending from its junction with the A92 Trunk Road to a point 40 metres or thereby north of the north kerbline of Barclayhill Place, other than in a southerly direction. The Order will also prohibit any vehicle from entering the gap in the central reservation of the A92 Trunk Road opposite Cookston Road, Portlethen.

Full details of the proposals are in the draft Order which, together with a map showing the lengths of the road affected and a statement of the Council's reasons for proposing to make the Order, may be examined during normal office hours in the Roads Department, Woodhill House, Ashgrove Road West, Aberdeen, and at the offices of the Divisional Road Surveyor, Kincardine and Deeside Divisional Office, Westfield House, Arduthie Road, Stonehaven.

Any person wishing to object to the proposed Order should send details of the grounds of objection in writing to the undersigned by 22nd August 1986.

Alan G. Campbell, Director of Law and Administration.

Woodhill House, Ashgrove Road West, Aberdeen.

### **GRAMPIAN REGIONAL COUNCIL**

# **ROAD TRAFFIC REGULATION ACT 1984**

# THE GRAMPIAN REGIONAL COUNCIL (HILL OF RUBISLAW, ABERDEEN)

# (TRAFFIC MANAGEMENT) ORDER 1986

THE Grampian Regional Council has made the above-named Order under Sections 1(1) and (2), and 2(1) to (3), of the Road Traffic Regulation Act 1984.

The effect of the Order, which will come into operation on 11th August 1986, will be to prohibit any vehicle on any of the roads specified in Column 1 of the First Schedule to this Notice, from making a righthand turn into the road specified opposite to that road in Column 2 of the said first Schedule and to prohibit any vehicle on any of the roads specified in Column 1 of the Second Schedule to this notice from entering the road specified opposite to that road in Column 2 of the said Second Schedule. Members of the public are asked to note that the Order will be implemented in stages with the restriction at Earl's Court Gardens coming into operation first, followed by Carnegie Crescent and King's Cross Road and, finally, Rubislaw Den Gardens, Royfold Crescent and Rubislaw Den South.

Copies of the Order and the relevant map are available for inspection during normal working hours at the offices of the Director of Law and Administration, Grampian Regional Council, Woodhill House, Ashgrove Road West, Aberdeen.

Any person who wishes to question the validity of the Order or of any provision contained within the Order on the grounds that it is not within the relevant section of the Act, or that any of the relevant requirements arising from the Act have not been complied with in relation to the Order may, within six weeks of 28th July 1986, apply to the Court of Session for this purpose.

### FIRST SCHEDULE

### Column 1

**Rubislaw Den South** 

Anderson Drive

Anderson Drive

Anderson Drive

Anderson Drive

Anderson Drive

Carnegie Crescent

King's Cross Road

Rubislaw Den Gardens

Earl's Court Gardens

Column 2 Anderson Drive Royfold Crescent Carnegie Crescent King's Cross Road Rubislaw Den Gardens Earl's Court Gardens

Anderson Drive

Anderson Drive

Anderson Drive

Anderson Drive

Director of Law and

Administration.

### SECOND SCHEDULE

Column 1	Column 2
Anderson Drive Royfold Crescent	Rubislaw Den South Anderson Drive
	Alan G. Campbell,

Woodhill House, Ashgrove Road West, Aberdeen AB9 2LU.

(22)

# NOTICE OF INTENTION TO PREPARE AN UPDATE OF THE STRATHCLYDE STRUCTURE PLAN

# TOWN AND COUNTRY PLANNING (SCOTLAND) ACT 1972

THE Strathclyde Regional Council intend to prepare an update of the Strathclyde Structure Plan to consider issues arising with regard to:-

- (a) Private Housing
- (b) The Greenbelt
- (c) Strategic Industrial Issues
- (d) Shopping
- (e) Mineral Aggregates

The Council invites organisations and individuals to indicate their interest in being consulted on the Update, by Friday, 15th August 1986.

Responses should be addressed to me at the undernoted address and marked "Structure Plan Update".

Robert Calderwood, Chief Executive.

Strathclyde Regional Council, 20 India Street, Glasgow G2 4PF. 25th July 1986.

(15)

# ROADS (SCOTLAND) ACT 1984

# THE STRATHCLYDE REGIONAL COUNCIL (M80 STEPPS BYPASS) (PROVAN INTERCHANGE TO THE CITY OF GLASGOW DISTRICT BOUNDARY) SPECIAL ROAD SCHEME 1984 AND THE STRATHCLYDE REGIONAL COUNCIL (M80 STEPPS BYPASS) (PROVAN INTERCHANGE TO CITY OF GLASGOW DISTRICT BOUNDARY) (CONNECTING ROADS) SPECIAL ROAD SCHEME 1984

THE Secretary of State hereby gives notice that he has confirmed the following Schemes.

- (i) The Strathclyde Regional Council (M80 Stepps Bypass) (Provan Interchange to City of Glasgow District Boundary) Special Road Scheme 1984 authorising the provision of a Special Road (Motorway) from the M8 Monkland Motorway to a point on the City of Glasgow District Boundary 100 metres north and 170 metres east of Saughs Farm, near Millerston and
- (ii) The Strathclyde Regional Council (M80 Stepps Bypass) (Provan Interchange to City of Glasgow District Boundary) (Connecting Roads) Special Road Scheme 1984 authorising the provision of connecting roads to the eastbound and westbound carriageways of the proposed Special Road (M80 Stepps Bypass).

Copies of the Confirmation Instruments which are entitled the Strathclyde Regional Council (M80 Stepps Bypass) (Provan Interchange to City of Glasgow District Boundary) Special Road Scheme 1984 Confirmation Instrument 1986 and the Strathclyde Regional Council (M80 Stepps Bypass) (Provan Interchange to City of Glasgow District Boundary) (Connecting Roads) Special Road Scheme 1984 Confirmation Instrument 1986 and each of which contains a copy of the scheme as confirmed may be inspected, free of charge, at all reasonable hours at the offices of the Strathclyde Regional Council, Melrose House, 19 Cadogan Street, Glasgow and the Scottish Development Department, New St Andrew's House, Edinburgh.

Any person aggrieved by the Schemes who desires to question the validity thereof or of any provision sustained therein, on the ground that it is not within the powers of the Roads (Scotland) Act 1984 or that any requirement of that Act has not been complied with in relation to the Scheme, may within six weeks from 1st August 1986 make application to the Court of Session for the suspension of the operation of, or opposing of, the Scheme or of any provision contained therein.

M.J.P. Cunliffe, Assistant Secretary, Scottish Office.

Scottish Development Department, New St Andrew's House, Edinburgh. 16th July 1986.

# THE STRATHCLYDE REGIONAL COUNCIL INDIA STREET, ALEXANDRIA PROPOSED ROAD CLOSURE

THE Strathclyde Regional Council propose to consider the introduction of a Traffic Regulation Order closing India Street at its junction with the Lomond Industrial Estate access road.

Full details of these proposals, which form the proposed The Strathclyde Regional Council (India Street, Alexandria) (Prohibition of Entry) Order 198, together with a plan and a statement of reasons for proposing to make this Traffic Regulation Order, are available for inspection during normal office hours Monday to Friday inclusive at the offices of:-

- (i) The Chief Executive, Strathclyde House 20 India Street, Glasgow G2 4PF
- (ii) The Divisional Engineer, Department of Roads Regional Offices, Garshake Road, Dumbarton (telephone enquiries to Dumbarton 65151 extension 115)
- (iii) The Area Officer, Chief Executive's Department Gilmour Street, Alexandria G83 7AB.

Any person wishing to object to the proposals should send details of the grounds for objection in writing to the Chief Executive, Strathclyde House, 20 India Street, Glasgow G2 4PF, by 22nd August 1986.

> Robert Calderwood, Chief Executive.

> > (9)

Strathclyde House, 20 India Street, Glasgow G2 4PF.

### THE STRATHCLYDE REGIONAL COUNCIL

### (BUCHANAN STREET/GRAHAM STREET, AIRDRIE)

### (TRAFFIC MANAGEMENT) ORDER 1986

ON 22nd July 1986 the Regional Council made the above named Order under Sections 1(1), (2), 2(1) to (3) of the Road Traffic Regulation Act 1984.

This Order, which comes into operation on Sunday, 10th August 1986 re-opens the present closure of Buchanan Street at Graham Street and introduces one-way operation southwards in Buchanan Street and westwards in Graham Street between Buchanan Street and Airdrie New Cross.

A copy of the Order and a Map relative thereto are available for inspection during normal working hours at the offices of:-

- (i) The Chief Executive, Strathclyde House,
  - 20 India Street, Glasgow,
- (ii) The Divisional Roads Engineer, Strathclyde Regional Offices, Almada Street, Hamilton (Telephone enquiries to Hamilton 282828 ext 344).

Any person who wishes to question the validity of the Order or of any provision contained in it on the grounds that it is not within the powers conferred by the Road Traffic Regulation Act 1984 or on the grounds that any requirement of that Act or of any instrument made under it has not been complied with in relation to the Order, may within six weeks from 1st August 1986, apply to the Court of Session for this purpose.

> Robert Calderwood, Chief Executive.

Strathclyde House, 20 India Street, Glasgow.

(3)

### ALANEX FLOORING LIMITED

NOTICE is hereby given that in a Petition by the Lord Advocate as representing Her Majesty's Commissioners of Customs & Excise craving the Court to order that Alanex Flooring Limited, a Company incorporated under the Companies Acts and having its Registered Office at 116 High Street, Irvine, Ayrshire, be wound up by the Court and to appoint an Official Liquidator of the said Company, the Sheriff at Kilmarnock, by Interlocutor, dated 22nd July 1986, ordered all parties interested if they intend to show cause why the Prayer of the Petition should not be granted, to lodge Answers in the hands of the Sheriff Clerk, Sheriff Court House, St Marnock Street, Kilmarnock, within eight days after intimation, service or advertisement.

> Shepherd & Wedderburn, W.S., Agent for the Petitioner.

17 Charlotte Square, Edinburgh.

(56)

### AMSY LIMITED

### (In Liquidation)

NOTICE is hereby given that following upon a Note being presented to the Sheriff of Grampian, Highland and Islands at Aberdeen by Ian Patrick Soutar, Chartered Accountant, Braemar House, 267 Union Street, Aberdeen, Official Liquidator of Amsy Limited, a Company incorporated under the Companies Acts, and having its Registered Office at 7 Bon Accord Crescent, Aberdeen, the Sheriff, by Interlocutor, dated 28th July (1) dispensed with the appointment of a Committee of Inspection; (2) dispensed with the settlement of a list of contributories; (3) fixed 1st September 1986 as a date on or before which all Creditors of the Company are to prove their debts or claims or to be excluded from the benefits of any distribution; (4) confirmed the appointment of Shepherd & Wedderburn, WS, 16 Charlotte Square, Edinburgh, as Solicitors to assist the Official Liquidator in the performance of his duties. All of which notice is hereby given.

Shepherd & Wedderburn, WS.

16 Charlotte Square, Edinburgh.

(58)

### ATLANTIC INSTRUMENTS LIMITED

# (In Liquidation)

### Winding Up by Order of the Court

### Registered Office : 2 Queens Road, Aberdeen

NOTICE is hereby given, pursuant to Section 546(2) of the Companies Act 1985 that a Meeting of Creditors of the above Company will be held within the offices of Scott-Moncrieff, Thomson & Shiells, C.A., 17 Melville Street, Edinburgh, on Wednesday, 3rd September 1986, at 10.30 a.m., for the purpose of determining whether or not an application is to be made to the Court for the appointment of a Committee of Inspection to act with the Liquidator, and who are to be Members of the Committee, if appointed.

> J.H. Strickland, Official Liquidator.

17 Melville Street, Edinburgh.

(31)

### BURNSIDE GARAGE TRANSPORT LIMITED

### (In Liquidation)

NOTICE is hereby given that in a Petition presented to the Sheriff of the Sheriffdom of Lothian and Borders at Linlithgow, the Sheriff, by Interlocutor, dated 23rd July 1986, ordered that Burnside Garage (57)

Transport Limited, a Company incorporated under the Companies Acts and having its Registered Office at Glendevon, Winchburgh, West Lothian, be wound up by the Court under provisions of the Companies Act 1985 and that Peter Cranbourne Taylor, Chartered Accountant, 29 Abercromby Place, Edinburgh, be appointed Official Liquidator of the said Company.

Shepherd & Wedderburn, W.S.

16 Charlotte Square, Edinburgh.

### **COLEY & STEWART LIMITED**

### (In Liquidation)

### 591 Nitshill Road, Darnley, Glasgow

NOTICE is hereby given pursuant to Section 546(2) of the Companies Act 1985, that a Meeting of Creditors of the above Company will be held within 100 Wellington Street, Glasgow G2 6DJ, on Wednesday, 20th August 1986, at 11 a.m., for the purposes of considering whether or not application is to be made to the Court for the appointment of a Committee of Inspection to act with the Official Liquidator and who are to be the Members of the Committee, if appointed. All persons having claims against the Company are requested to lodge these with the Official Liquidator forthwith and persons indebted to the Company are requested to make payment to him without delay.

> G. Ian Rankin, C.A., Official Liquidator.

100 Wellington Street, Glasgow G2 6DJ.

29th July 1986.

# (29)

### DOUGLASFLOOR LIMITED

### In Liquidation

A NOTE having been presented to the Sheriff at Dumbarton by the Official Liquidator of the above Company, Leon Marshall, C.A., 24 Sandyford Place, Glasgow, the Sheriff, by Interlocutor, dated 28th July 1986, (1) sanctioned the appointment of Bird Semple & Crawford Herron, Solicitors, 249 West George Street, Glasgow, as Solicitors in the Liquidation to assist the Liquidator in the performance of his duties, (2) fixed 26th August 1986 as the date on or before which Creditors must prove their debts or claims or be excluded from any distribution made before such proof.

Of which intimation is hereby given.

Bird Semple & Crawford Herron, Solicitors.

249 West George Street, Glasgow G2 4RB.	
Official Liquidator's Agents.	(75)

### DUNFERMLINE GOLF CENTRE LTD

# (In Liquidation)

NOTICE is hereby given pursuant to Section 546 of the Companies Act 1985 that a Meeting of Creditors of the above Company will be held in the Oak Room, 3 Randolph Crescent, Edinburgh, on Friday, 29th August 1986, at 10.30 a.m., for the purposes of determining whether or not an application is to be made to the Court for the appointment of a Committee of Inspection to act with the Liquidator, and who are to be Members of the Committee if appointed.

> T. Ritchie Campbell, C.A., Official Liquidator.

T Hunter Thomson & Co., 28 Alva Street, Edinburgh EH2 4QF.

29th July 1986.

## WILLIAM FENWICK & SON (DUNDEE) LIMITED

### (In Members' Voluntary Liquidation)

NOTICE is hereby given, in pursuance of Section 290 and 341(1)(b) of the Companies Act 1948, that a General Meeting of the above named Company will be held at 144 Nethergate, Dundee, on Thursday, 18th September, at 10.30 a.m., for the purpose of having an account laid before the Members, showing the manner in which the winding up has been conducted, and the property of the Company disposed of, of hearing any explanation that may be given by the Liquidator, and also of determining, by Extraordinary Resolution, the manner in which the books, accounts and documents of the Company and of the Liquidator shall be disposed of.

> Murray G. Dalgety, Liquidator.

24th July 1986.

(59)

Notice to Members of Final Meeting

(Members' Voluntary Winding Up)

Pursuant to Sections 585 and 640(1) of the Companies Act 1985

### McEWAN'S UTILITY HOUSE LIMITED

### (In Voluntary Liquidation)

NOTICE is hereby given, pursuant to Sections 585 and 640(1)(b) of the Companies Act 1985, that a General Meeting of the Members of the above-named Company will be held at Scottish Provident House, 52 Brown Street, Manchester M2 2AU, on Monday, 8th September 1986, at 3 p.m. precisely, for the purpose of having an account laid before them, and to receive the Liquidators report, showing how the winding up of the Company has been conducted, and its property disposed of, and of hearing any explanation that may be given by the Liquidator; and also of determining by Extraordinary Resolution the manner in which the books, accounts, papers and documents of the Company and of the Liquidator thereof, shall be disposed of.

Any Member entitled to attend and vote at the above-mentioned Meeting is entitled to appoint a proxy to attend and vote instead of him, and such proxy need not also be a Member.

> D. M. Lewis, Liquidator.

Dated 29th July 1986.

(80)

### McINTOSH BROTHERS LIMITED

NOTICE is hereby given that in a Petition at the instance of McIntosh Brothers Limited to the Sheriff of Tayside, Central and Fife at Kirkcaldy, the Sheriff, by Interlocutor, dated 23rd July 1986, ordered that McIntosh Brothers Limited, 274 Links Street, Kirkcaldy, be wound up by the Court under the provisions of the Companies Act 1985, and that Alan Clark Thomson, C.A., 18 Viewfield Terrace, Dunfermline, be appointed Official Liquidator of the said Company.

> Ross & Connel, Solicitors.

10 Viewfield Terrace, Dunfermline.

Petitioner's Agents.

(32)

### MICRO PULSE DISPLAY LIMITED

### (In Liquidation)

INTIMATION is hereby given that by Interlocutor, dated 23rd July 1986, the Sheriff at Cupar Sheriff Court appointed me Official Liquidator of Micro Pulse Display Limited.

Notice is hereby given, pursuant to Section 546 of the Companies Act 1985 that a Meeting of Creditors of the said Micro Pulse Display Limited will be held within the Roxburghe Hotel, 38 Charlotte Square,

(54)

Edinburgh, on Wednesday, 27th August 1986, at 12 noon, for the purpose of determining whether or not application is to be made to the Court for the appointment of a Committee of Inspection to act with the Liquidator and who are to be the Members of the Committee if appointed.

Murdoch L. McKillop, Official Liquidator.

Arthur Andersen & Co., 199 St. Vincent Street, Glasgow G2 5QD.

29th July 1986.

(24)

# MONT DEVELOPMENTS LIMITED

NOTICE is hereby given that in a Petition by Mont Developments Limited, having their Registered Office at 16 Broomvale Drive, Newton Mearns, Glasgow, craving the Court *inter alia* to order that the said Mont Developments Limited, incorporated under the Companies Acts and having their Registered Office at 16 Broomvale Drive, Newton Mearns, Glasgow, be wound up by the Court and to appoint an Official Liquidator of the said Company, the Sheriff of North Strathclyde at Paisley, by Interlocutor, dated 4th July 1986, ordained all parties desirous of opposing to lodge Answers in the hands of the Sheriff Clerk at Paisley within eight days after intimation, advertisement and service.

> Dorman, Jeffrey & Co., Solicitors.

Provincial House, 140 West George Street, Glasgow G2 2HH.

Agent for Petitioners.

(26)

(25)

Companies Form No. 600a

Notice of Appointment of Liquidator

Creditors Voluntary Winding Up

Pursuant to Section 600 of the Companies Act 1985

Name of Company	NORTH SEA ENGINEERING LIMITED
Previous names of Company	Dustink Limited North Sea Sales Limited
Nature of Business	Engineering service to the Oil and Gas Industry
Address of Registered Office	135 Wellington Street, Glasgow G2 2XE
Liquidator's Name and Address	Michael D. Sheppard, Chartered Accountant, 135 Wellington Street, Glasgow G2 2XE
Date of Appointment	16th July 1986
By Whom Appointed	Creditors of the Company
	Michael D. Sheppard, Liquidator.

Date 28th July 1986.

### NOVONYMOUS LIMITED

### (In Members' Voluntary Liquidation)

NOTICE is hereby given, in terms of Section 585, Companies Act 1985, that an Extraordinary General Meeting of the Members of the above Company will be held at 25 Charlotte Square, Edinburgh, on Monday, 8th September 1986, at 10 a.m., to receive the Liquidator's account and report on the winding up; and also to determine by Extraordinary Resolution the manner in which the books, accounts, papers and documents of the Company and of the Liquidator thereof, shall be disposed of.

*Note:* A Member entitled to attend and vote at the above Meeting is entitled to appoint a proxy, who need not be a Member of the Company, to attend the Meeting and vote on his behalf on a poll.

D. K. McLellan, Liquidator.

25 Charlotte Square, Edinburgh EH2 4EZ. Dated 30th July 1986.

(55)

### ORTLICH LIMITED

### (In Provisional Liquidation)

NOTICE is hereby given that in a Petition by Scottish Development Agency, having their principal place of business at 120 Bothwell Street, Glasgow, craving the Court *inter alia* to order that Ortlich Limited, incorporated under the Companies Acts, and having their Registered Office at 10 Eastmuir Street, Annick Street Development, Shettleston, Glasgow, be wound up by the Court and to appoint an Official Liquidator of the said Company, the Sheriff at Glasgow, by Interlocutor, dated 17th July 1986, ordered all parties desirous of opposing to lodge Answers in the hands of the Sheriff Clerk at Glasgow within eight days after intimation, advertisement or service, and in the meantime, until the Prayer of the Petition has been granted or refused, nominated Raymond Ellis Blin, C.A., of 40 Wellington Street, Glasgow, to be Provisional Liquidator of the said Company on his finding Caution before Extract.

> Mitchells Roberton, Petitioners' Agent.

George House, 36 North Hanover Street, Glasgow G1 2AD.

(63)

### PASDENOM LIMITED

### (In Members' Voluntary Liquidation)

NOTICE is hereby given in terms of Section 595 of the Companies Act 1985 that the Final Meeting of Members of the Company will be held at 18 Alva Street, Edinburgh EH2 4QN, on Tuesday, 2nd September 1986, at 12 noon, for the purpose of receiving the Liquidator's report showing how the winding up of the Company has been conducted, to hear any explanations which may be given by the Liquidator, and to determine the manner in which the books and documents of the Company and of the Liquidator shall be disposed of.

> James Dean, C.A., Liquidator.

Smith Hemming & Dean, C.A., 18 Alva Street, Edinburgh EH2 4QN. 30th July 1986.

.

(51)

### PAULISA LIMITED

### (In Liquidation)

NOTICE is hereby given that following a Note presented to the Sheriff of Glasgow and Strathkelvin at Glasgow by Kenneth Alexander Ross, Chartered Accountant, 77 Renfrew Street, Glasgow, the Official Liquidator of Paulisa Limited, a Company incorporated under the Companies Acts and having its Registered Office at 104 West George Street, Glasgow, the Sheriff at Glasgow, by Interlocutor, dated 23rd July 1986 (1) dispensed with the settlement of a list of contributories; (2) dispensed with the appointment of a Committee of Inspection; (3) sanctioned and confirmed the appointment of Shepherd & Wedderburn, WS, 16 Charlotte Square, Edinburgh, as Solicitors to assist the Official Liquidator in the performance of his duties; (4) fixed 29th August 1986 as a date on or before which all Creditors of Paulisa Limited are to

Name of Company

present debts or claims or be excluded from the benefits of any distribution made before the said debts are proved and appointed that a copy of the Note and Interlocutor be posted on the Walls of the Sheriff Court, Glasgow, and notice of the import of the said Note and Deliverance to be advertised in the Edinburgh Gazette and Glasgow Herald newspapers. All of which notice is hereby given.

> Shepherd & Wedderburn, W.S., Agents for the Noter.

> > (47)

(82)

(76)

(17)

### WILLIAM PEACOCK LIMITED

#### Cable Works, Chain Road, Paisley

NOTICE is hereby given in pursuance of Section 588 of the Companies Act 1985 that a Meeting of the Creditors of the above will be held within the Merchants House, 7 West George Street, Glasgow, on Tuesday, 12th August 1986, at 12 noon, for the purposes specified in Sections 588 to 590 of the said Act.

By Order of the Board.

W. A. Peacock, Director.

Registered Office: Cable Works, Chain Road, Paisley.

Dated 29th July 1986.

### DOUGLAS REID (BUILDING SERVICES) LIMITED

#### In Liquidation

A NOTE having been presented to the Sheriff at Dumbarton by the Official Liquidator of the above Company, Leon Marshall, C.A., 24 Sandyford Place, Glasgow, the Sheriff, by Interlocutor, dated 28th July 1986, (1) sanctioned the appointment of Bird Semple & Crawford Herron, Solicitors, 249 West George Street, Glasgow, as Solicitors in the Liquidation to assist the Liquidator in the performance of his duties, (2) fixed 26th August 1986 as the date on or before which Creditors must prove their debts or claims or be excluded from any distribution made before such proof.

Of which intimation is hereby given.

Bird Semple & Crawford Herron, Solicitors.

249 West George Street, Glasgow G2 4RB.

Official Liquidator's Agents.

### SMITH & COULSON (IRONMONGERS) LIMITED

NOTICE is hereby given that by Interlocutor of the Sheriff of Tayside Central and Fife at Stirling, dated 22nd July 1986, John McAulay, Chartered Accountant, 64 Murray Place, Stirling, was appointed Official Liquidator of Smith & Coulson (Ironmongers) Limited.

> Kerr & Co., Solicitors.

20 Murray Place, Stirling.

Agents for Official Liquidator.

# Notice to Members of Final Meeting

### (Members' Voluntary Winding Up)

Pursuant to Sections 585 and 640(1) of the Companies Act 1985

### SMITH'S WAREHOUSE (GREENOCK) LIMITED

### (In Voluntary Liquidation)

NOTICE is hereby given, pursuant to Sections 585 and 640(1)(b) of the Companies Act 1985, that a General Meeting of the Members of the above-named Company will be held at Scottish Provident House, 52 Brown Street, Manchester M2 2AU, on Monday, 8th September 1986, at 4 p.m. precisely, for the purpose of having an account laid before them, and to receive the Liquidators report, showing how the winding up of the Company has been conducted, and its property disposed of, and of hearing any explanation that may be given by the Liquidator; and also of determining by Extraordinary Resolution the manner in which the books, accounts, papers and documents of the Company and of the Liquidator thereof, shall be disposed of.

Any Member entitled to attend and vote at the above-mentioned Meeting is entitled to appoint a proxy to attend and vote instead of him, and such proxy need not also be a Member.

> D. M. Lewis. Liquidator.

Dated 29th July 1986.

(81)

### Special Resolution

### of

### STRATHCLYDE ELECTRONICS LIMITED

Passed

AT an Extraordinary General Meeting of the above Company, duly convened and held at 256 West George Street, Glasgow, on 25th July 1986, the subjoined Special Resolution was duly passed viz:

"That the Company be wound up voluntarily, and that Robert Boyd Copleton, C.A., of John M. Taylor and Company, C.A., be hereby appointed Liquidator for the purposes of such winding up."

> R. R. Hammond, Director.

> > (60)

### The Companies Act 1985

Members' Voluntary Winding Up

### Notice of Appointment of Liquidator

STRATHCI VDE

# Pursuant to Section 600

Nume of Company	ELECTRONICS LIMITED
Nature of Business	Service and Maintenance of Electronic Equipment
Address of Registered Office	105 Clarence Drive, Glasgow
Liquidator's Name and Address	Robert Boyd Copleton, C.A., John M. Taylor and Company, C.A., 256 West George Street, Glasgow G2 4QY
Date of Appointment	25th July 1986
By Whom Appointed	The Members of the Company
	Robert P. Condition C. 4

Robert B. Copleton, C.A., Liquidator.

1179



### THE EDINBURGH GAZETTE 1st AUGUST 1986

### TAYINLOAN SALMON LIMITED

### (In Creditors' Voluntary Liquidation)

### Formerly trading from Rhunahaorine Point, Tayinloan, Argyll

IN pursuance of Section 595 Companies Act 1985, a Meeting of the Members will be held on Friday, 12th September 1986, at 3 p.m., within 29 Abercromby Place, Edinburgh 3, and a Meeting of the Creditors will be held on the same day and at the same place at 3.30 p.m., for the purpose of receiving the Liquidator's final report showing how the winding up has been conducted, and of hearing any explanations which may be given by the Liquidator.

> Peter C. Taylor, C.A., Liquidator.

29 Abercromby Place, Edinburgh EH3 6UE. 30th July 1986.

(83)

(52)

### VALUEGLAZE LIMITED

### In Liquidation

NOTICE is hereby given that a Meeting of the Creditors of the above Company will be held within the Chambers of Cork Gully, Chartered Accountants, 76 George Street, Edinburgh, on 7th August 1986, at 11 a.m., for the purposes of receiving an account of the Liquidator's acts and dealings and of the conduct of the winding up during the year ended 7th May 1986.

> R B M Graham, Liquidator.

Cork Gully, 76 George Street, Edinburgh EH2 3BU. .30th July 1986.

# WAVERLEY BUILDING SERVICES LIMITED

### (In Liquidation)

NOTICE is hereby given that on 24th July 1986 a Note was presented to the Sheriff of South Strathclyde, Dumfries and Galloway at Airdrie, by Roy Arthur Johnson, Chartered Accountant, Kintyre House, 209 West George Street, Glasgow, the Official Liquidator of Waverley Building Services Limited (In Liquidation), incorporated under the Companies Acts and having its Registered Office at 52 Telford Road, East Lenziemill Industrial Estate, Cumbernauld, for examination and audit of his accounts, fixing of his remuneration, approval of his intromissions, taxation of his Solicitor's accounts, authorisation of payments, examination and discharge of him as Official Liquidator and the dissolution of the Company. By Deliverance, dated 24th July 1986, the said Sheriff ordered intimation on the Walls of Airdrie Sheriff Court, service on Commercial Union Assurance plc, the Official Liquidator's Cautioners and the advertisement in the Glasgow Herald and Edinburgh Gazette, and ordained all parties desirous of opposing the Note to lodge Answers with the Sheriff Clerk at Airdrie within eight days after such intimation, service and advertisement, under certification.

> Bishop & Co., Solicitors.

129 St. Vincent Street, Glasgow. Agents for the Noter.

(46)

Sequestration of the Estates of

### A & B INSURANCE BROKERS

### (formerly trading at 580 George Street, Aberdeen)

THE Estates of A & B Insurance Brokers, formerly trading from 580 George Street, Aberdeen, and Alexander Buchan, 4 Brook Crescent, Denmore Park, Bridge of Don, Aberdeen, and Bruce Mackenzie Sneddon, 7 Firthview Road, Muirton, Inverness, the Partners of the said Firm, were Sequestrated by the Sheriff of Grampian Highland and Islands at Aberdeen, on 23rd July 1986, and Oliver Lawson Balfour, Chartered Accountant, 6 Golden Square, Aberdeen, has been appointed by the Court to act as Interim Trustee on the Sequestrated Estates.

Any Creditor of the Debtors named above is invited to submit his statement of claim in the prescribed form, with any supporting accounts or vouchers, to the Interim Trustee.

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a Permanent Trustee.

Oliver L. Balfour, C.A., Interim Trustee.

Deloitte Haskins & Sells, 6 Golden Square, Aberdeen.

29th July 1986.

(65)

# Sequestration of

# MRS A. BORLAND

### 10 Carmichael Place, Langside, Glasgow

THE Estate of Mrs A. Borland, 10 Carmichael Place, Langside, Glasgow, was Sequestrated by the Sheriff of Glasgow and Strathkelvin at Glasgow on 16th July 1986, and Douglas B. Jackson, C.A., F.I.P.A., Dalziel House, 7 Claremont Terrace, Glasgow G3 7XR, has been appointed by the Court to act as Interim Trustee on the Sequestrated Estate.

Any Creditor of the Debtor named above is invited to submit his statement of claim in the prescribed form, with any supporting accounts or vouchers, to the Interim Trustee.

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a Permanent Trustee.

> Douglas B. Jackson, C.A., F.I.P.A., Interim Trustee.

29th July 1986.

(62)

### Sequestration of

### JAMES CRAIG

1 Castlehill Road, Stewarton, Ayrshire

# 52 Lainshaw Street, Stewarton, Ayrshire

THE Adjourned Public Examination of the Bankrupt will take place within the Sheriff Courthouse, Kilmarnock, on Thursday, 7th August 1986, at 10 a.m.

> Frank H. Mycroft, C.A., Trustee.

100 Wellington Street, Glasgow G2 6DJ. 29th July 1986.

(66)

### Bankruptcy (Scotland) Act 1985 : Section 15(6)

### Sequestration of the Estate of

### GEORGE FERGUSON

THE Estate of George Ferguson, 164 North High Street, Musselburgh, was Sequestrated by the Sheriff of Lothian and Borders at Haddington, on 21st July 1986, and Peter Leslie Stewart, C.A., 33/34 Charlotte Square, Edinburgh EH2 4HF, has been appointed by the Court to act as Interim Trustee on the Sequestrated Estate.

Any Creditor of the Debtor named above is invited to submit his statement of claim in the prescribed form, with any supporting accounts or vouchers, to the Interim Trustee.

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a Permanent Trustee.

Peter Leslie Stewart, Interim Trustee.

(79)

(86)

KMG Thomson McLintock, 33/34 Charlotte Square, Edinburgh EH2 4HF.

Date 30th July 1986.

#### Sequestration of

### HENRY ELLIS HASTON

### Residing at 55 Telford Drive, Edinburgh

THE Estate of Henry Ellis Haston, residing at 55 Telford Drive, Edinburgh, was Sequestrated by the Sheriff of Lothian & Borders at Edinburgh on 23rd July 1986, and P. Hamish Armour, Chartered Accountant, 29 Abercromby Place, Edinburgh, has been appointed by the Court to act as Interim Trustee on the Sequestrated Estate.

Any Creditor of the Debtor named above is invited to submit his statement of claim in the prescribed form, with any supporting accounts or vouchers, to the Interim Trustee.

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a Permanent Trustee.

> P. Hamish Armour, Interim Trustee.

1st August 1986.

Bankruptcy (Scotland) Act 1985 : Section 15(6)

### Sequestration of the Estate of

### MOHAMMED JAMIL AND AKHTAR BIBI JAMIL

THE Estate of Mohammed Jamil and Akhtar Bibi Jamil, The Post Office House, Westfield, Bathgate, was Sequestrated by the Sheriff of Lothian and Borders at Linlithgow, on 28th July 1986, and Raymond Ellis Blin, C.A., Pannell Kerr Forster, 40 Wellington Street, Glasgow G2 6RL, has been appointed by the Court to act as Interim Trustee on the Sequestrated Estate.

Any Creditor of the Debtor named above is invited to submit his statement of claim in the prescribed form, with any supporting accounts or vouchers, to the Interim Trustee.

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a Permanent Trustee.

> Raymond E. Blin, Interim Trustee.

Date 30th July 1986.

(67)

### Bankruptcy (Scotland) Act 1985 : Section 15(6)

Sequestration of the Estate of

#### ROGER JOHN LEADBEATER

THE Estate of Roger John Leadbeater, Hatton House, Chirnside, was Sequestrated by the Sheriff of Lothian and Borders, at Duns, on 23rd July 1986, and Peter Leslie Stewart, C.A., 33/34 Charlotte Square, Edinburgh EH2 4HF, has been appointed by the Court to act as Interim Trustee on the Sequestrated Estate.

Any Creditor of the Debtor named above is invited to submit his statement of claim in the prescribed form, with any supporting accounts or vouchers, to the Interim Trustee.

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a Permanent Trustee.

> Peter Leslie Stewart, Interim Trustee.

KMG Thomson McLintock, 33/34 Charlotte Square, Edinburgh EH2 4HF.

Date 30th July 1986.

(78)

### Sequestration of

### **RICHARD MAIN**

#### Residing at 35/6 Easter Drylaw Place, Edinburgh

THE Estate of Richard Main, residing at 35/6 Easter Drylaw Place, Edinburgh, was Sequestrated by the Sheriff of Lothian & Borders at Edinburgh on 23rd July 1986, and P. Hamish Armour, Chartered Accountant, 29 Abercromby Place, Edinburgh, has been appointed by the Court to act as Interim Trustee on the Sequestrated Estate.

Any Creditor of the Debtor named above is invited to submit his statement of claim in the prescribed form, with any supporting accounts or vouchers, to the Interim Trustee.

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a Permanent Trustee.

> P. Hamish Armour, Interim Trustee.

1st August 1986.

Date 29th July 1986.

(87)

(28)

Bankruptcy (Scotland) Act 1985 : Section 15(6)

### Sequestration of the Estate of

### DOUGLAS McDONALD

THE Estate of Douglas McDonald, formerly trading as McDonald Off-Sales, Kelso Street, Yoker, Glasgow, currently residing at 23 Lyle Square, Milngavie, was Sequestrated by the Sheriff of North Strathclyde at Dumbarton on 25th July 1986, and Raymond E. Blin, C.A., 40 Wellington Street, Glasgow G2 6RL, has been appointed by the Court to act as Interim Trustee on the Sequestrated Estate.

Any Creditor of the Debtor named above is invited to submit his statement of claim in the prescribed form, with any supporting accounts or vouchers, to the Interim Trustee.

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a Permanent Trustee.

Raymond E. Blin, Interim Trustee. Bankruptcy (Scotland) Act 1985 : Section 15(6)

### Sequestration of the Estate of

### ARTHUR ROBERT McMASTER

THE Estate of Arthur Robert McMaster, currently residing care of 20 New Plymouth, East Kilbride, was Sequestrated by the Sheriff of South Strathclyde at Dumfries and Galloway at Hamilton, on 25th July 1986, and Raymond E. Blin, C.A., 40 Wellington Street, Glasgow G2 6RL, has been appointed by the Court to act as Interim Trustee on the Sequestrated Estate.

Any Creditor of the Debtor named above is invited to submit his statement of claim in the prescribed form, with any supporting accounts or vouchers, to the Interim Trustee.

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a Permanent Trustee.

Sequestration of the Estate of ARIF MINHAS

Bankruptcy (Scotland) Act 1985 : Section 15(6)

THE Estate of Arif Minhas, residing at 25 Bridge Street, Dollar, was

Sequestrated by the Sheriff at Alloa, on 1st July 1986, and D. J. Watt,

Chartered Accountant, 76 George Street, Edinburgh, has been

appointed by the Court to act as Interim Trustee on the Sequestrated

statement of claim in the prescribed form, with any supporting accounts

date, time and place of the Statutory Meeting of Creditors to elect a

Any Creditor of the Debtor named above is invited to submit his

Any Creditor known to the Interim Trustee will be notified of the

Raymond E. Blin, Interim Trustee.

D. J. Watt, C.A.,

Interim Trustee.

Date 29th July 1986.

or vouchers, to the Interim Trustee.

Permanent Trustee.

Cork Gully.

76 George Street.

Edinburgh EH2 3BU.

25th July 1986.

Estate.

30th July 1986.

Aberdeen.

(27)

(64)

(85)

#### Sequestration of

Sequestration of the Estate of

GRAEME J RAE (Formerly trading as Aberdeen Glazing Co)

THE Estate of Graeme J Rae, 1A Skene Place, Dyce, Aberdeen, was Sequestrated by the Sheriff of Grampian Highlands and Islands at

Aberdeen, on 23rd July 1986, and Oliver Lawson Balfour, Chartered

Accountant, 6 Golden Square, Aberdeen, has been appointed by the

statement of claim in the prescribed form, with any supporting accounts

Any Creditor of the Debtor named above is invited to submit his

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a

Court to act as Interim Trustee on the Sequestrated Estate.

or vouchers, to the Interim Trustee.

Permanent Trustee.

Deloitte Haskins & Sells, 6 Golden Square,

### BRYCE JOHN ROBERTSON

THE Estate of Bryce John Robertson, Esq, 28 Edmonstone Avenue, Danderhall, Midlothian, was Sequestrated by the Sheriff at Edinburgh on 23rd July 1986, and Peter C. Taylor, C.A., 29 Abercromby Place, Edinburgh EH3 6UE, has been appointed by the Court to act as Interim Trustee on the Sequestrated Estate.

Any Creditor of the Debtor named above is invited to submit his statement of claim in the prescribed form, with any supporting accounts or vouchers, to the Interim Trustee.

Any Creditor known to the Interim Trustee will be notified of the date, time and place of the Statutory Meeting of Creditors to elect a Permanent Trustee.

Peter C. Taylor, Interim Trustee.

O. L. Balfour, Interim Trustee.

29 Abercromby Place, Edinburgh EH3 6UE.

30th July 1986.

Sequestration of

### MESSRS R & J NEILSON

# and the Partners thereof

Lochgelly, Fife

THE Trustee hereby intimates that an account of his intromissions with the Funds of the Estate, brought down to 3rd July 1986, has been audited by the Commissioner, and that a Third Interim Dividend will be paid to Unsecured Creditors on and after 3rd September 1986, to those Creditors whose claims have been lodged and admitted.

> Peter C. Taylor, C.A., Trustee.

29 Abercromby Place, Edinburgh EH3 6UE. 31st July 1986. MRS JANET GAIRNS YOUNG OR REDPATH'S

### Judicial Factory

I, WILLIAM McNICOL BROWNLIE, Chartered Accountant, 77 Renfrew Street, Glasgow, Judicial Factor on the Trust Estate constituted by the Disposition, dated 30th April and recorded G.R. (Glasgow) 21st May 1892 by Robert Barbour's Trustees in favour of and in trust for behoof of *inter alios* the survivors and survivor of the late Miss Margaret Young of Mount Villa, Mount Vernon, Glasgow, and latterly of 591 Dayton Avenue, St. Paul, Minnesota, U.S.A., the late Miss Helen Campbell Young of Mount Villa, Mount Vernon, Glasgow, subsequently Mrs Helen Campbell Redpath of St. Paul, Minnesota, U.S.A., and the late Miss Janet Gairns Young of Mount Villa, Mount Vernon, Glasgow, subsequently Mrs Janet Gairns Redpath of 432 West Twentieth Avenue, San Mateo, California, U.S.A. (all daughters of the late Peter Young, House Factor in Glasgow) conform to Interlocutor of Lord Cowie, dated 10th December 1985, considering that I am about to seek authority to distribute the Trust Estate in my charge following upon the sale of the heritable property situated at 5/7½ Muirpark Street,

(84)

(30)

Glasgow, request all parties believing they have claims against, or an interest in, the said Trust Estate to lodge claims with me by 1st October 1986.

William McNicol Brownlie, Judicial Factor.

Ernst & Whinney, Savoy Tower, 77 Renfrew Street, Glasgow.

30th July 1986.

(74)

# Court Ref. No. B172/86

A PETITION has been raised in Selkirk Sheriff Court by William Laidlaw, Petitioner, for appointment of the Petitioner as *executor dative qua* funerator of the late JOHN RAE BISHOP LAIDLAW, formerly of 45 Glebe Place, Galashiels. Any persons desiring to oppose the appointment must lodge objections thereto in the hands of the Sheriff Clerk at Selkirk within fourteen days of the date of this Notice.

James Cullen, Solicitor.

Royal Bank Chambers, Melrose.

Solicitor for Pursuer.

(73)

# Hansard

# The Official Verbatim Report of Parliamentary Proceedings

Hansard reflects opinion on all topics of the day and is of great interest to the business or professional man. In addition to a factual word-for-word transcription of speeches in debate, it contains the answers to all Parliamentary Questions, complete Division Lists, and following each session details of the composition of the House. Proceedings up to 10 p.m. each day are reported in the issue on sale at 8.30 the following morning at the Government Bookshop in Holborn. A Weekly Edition is also published, and bound volumes are issued periodically during the session. Standing Orders can be accepted, if a suitable deposit is sent, for debates on definite subjects.

# SUBSCRIPTION RATES

House of Commons		House of Lords
2420	DAILY PARTS	
£430	Annual	£335
	WEEKLY HANSARD	
£240	Annual	£175
£55.35	Index only	£38.50

(Published Fortnightly)

Index to the House of Commons Debates per volume £5.00

All subscriptions rates are inclusive of postage.

# The London Gazette

# THE LONDON GAZETTE

Mondays, Tuesdays, Wednesdays, Thursdays and Fridays (except Good Friday and Public Holidays).

Single numbers £1.00 including postage. Annual subscription including postage £250.00.

# EDINBURGH GAZETTE

# AUTHORISED SCALE OF CHARGES FOR NOTICES AND ADVERTISEMENTS

Note; each notice is regarded as a separate advertisement

		0110F 5 0	ddrassas /roada
		5 addresses/roads	
	Conservation Areas — Local Plans — Stopping Up and Conversion of Roads	£40.25 (35.00	+ VAT) for up to
G.	Town and Country Planning (Scotland) Acts — Listed Buildings — Listed Buildings in		
F.	Friendly Societies	£13.23	(£11.50 + VAT)
E.	Sequestrations/Trust Deeds — all notices	£34.50	(£30.00 + VAT)
D.	Notice of Application for Winding Up by the Court	£34.50	(£30.00 + VAT)
С.	Notices to Creditors and Annual/Final Meetings of Members/Creditors	£28.75	(£25.00 + VAT)
B.	Notices of Resolutions and Meetings of Members/Creditors	£34.50	(£30.00 + VAT)
А.	Notice of appointment of Liquidator/Receiver	£28.75	(£25.00 + VAT)

over 5 addresses/roads £80.50 (£70.00 + VAT)

H. Notices outwith the above categories will be charged at a line rate of  $\pounds 21.85$  ( $\pounds 19.00 + VAT$ ) for up to 10 lines of printed matter and an additional £5.75 (£5.00 + VAT) for each 5 lines or part of 5 lines. The charges under A-D are in respect of notices covering one company only. If notices include

more than one company, in the same group of companies they will be charged as follows:-2 to 5 companies

double the single company rate 6 to 10 companies treble the single company rate

Pre-payment is required for these rates. In the case of notices under H. above pre-payment of £21.85 (£19.00 + VAT) is required. You will be invoiced for any balance.

All Notices and Advertisements are published in the Edinburgh Gazette at the risk of the Advertiser and at the discretion of the Editor. Whilst every endeavour will be made to ensure that the publication is made in accordance with the advertiser's requirements the Crown accepts no responsibility for any loss or damage howsoever arising from either a failure to meet those requirements or in respect of any errors or omissions which may inadvertently be made in respect of those requirements or in connection with any notice or advertisement.

Notices and Advertisements by Private Advertisers may be tendered at or sent directly by post to The Edinburgh Gazette, Her Majesty's Stationery Office, 13a Castle Street, Edinburgh EH2 3AR, for insertion at the authorised rates of payment. All notices and Advertisements must be prepaid. The Office hours are from 9.00 a.m. to 4.00 p.m. Mondays to Fridays, inclusive. Notices and Advertisements sent direct by post must be accompanied by postal orders or cheques made payable to The Edinburgh Gazette and should reach the office before 10.00 a.m. on the day prior to publication. Advertisements cannot be withdrawn after 10.00 a.m. on the day prior to publication. The dues on withdrawn Advertisements cannot be returned.

> A voucher copy will be posted on the day of publication on prepayment of 45p inclusive of postage. Edinburgh Gazette Notice charges include Value Added Tax at 15%.

> Published by HMSO at 13a Castle Street, Edinburgh EH2 3AR. Tel: 031 225 6333. Printed by John Swain & Son (Edinburgh) Ltd., 9 King's Haugh, Peffermill, Edinburgh EH16 5UY.

Edinburgh Gazette Annual Subscription £52.52. Single copy 45p

Company Law Official Notifications Supplement Annual Subscription £52.52.

Joint Annual Subscription for Gazette and Supplement £92.04.

ISBN 0 11 493187 9